10-26-1999 Docket No.: FORM PTO-1594 (Modified) **IEET** (Rev. 6-13) OMB No 0651-0011 (exp.4/94) Copy == 1994-97 LegalStar nwgc. vos Tab settings → → 101185009 Trademarks. Flease record the attached original documents or copy thereof. To the Honorable Commissioner of Pa 2. Name and address of receiving party(ies): 1. Name of conveying party(ies): **Bally Entertainment Corporation** Name: Hilton Hotels Corporation Internal Address: Street Address: 9336 Civic Center Drive Association ☐ Individual(s) Limited Partnership ☐ General Partnership City: Beverly Hills State: CA ZIP: 90210 ☒ Corporation-State Delaware ☐ Individual(s) citizenship ☐ Other ☐ Yes ☐ No Association Additional names(s) of conveying party(ies) ☐ General Partnership \_\_\_\_\_ 3. Nature of conveyance: ☐ Limited Partnership \_\_\_\_\_\_ ★ Corporation-State <u>California</u> Merger Assignment Change of Name ☐ Other \_\_\_\_ ☐ Security Agreement □ Other If assignee is not domiciled in the United States, a domestic ☐ Yes designation is Execution Date: June 6, 1996 (Designations must be a separate document from Additional name(s) & address(es) ☐ Yes  $\square$  N Application number(s) or registration numbers(s): B. Trademark Registration No.(s) A. Trademark Application No.(s) 1,773,050 ☐ Yes ☐ No Additional numbers 6. Total number of applications and 5. Name and address of party to whom correspondence registrations involved:.... concerning document should be mailed: Name: Julie A. Katz, Esq. 7. Total fee (37 CFR 3.41):....\$ \$40.00 Internal Address: Welsh & Katz, Ltd. ☑ Enclosed Authorized to be charged to deposit account 8. Deposit account number: Street Address: 120 South Riverside Plaza, 22nd Floor 23-0920 City: Chicago State: IL ZIP: 60606 DO NOT USE THIS SPACE 14/26/1999 NTHAI1 00000004 1773050 FC:481 40.00 BP 9. Statement and signature. To the best of my knowledge and belief, the foregoing information is true and correct and any attached copy is a true copy of the original document. 10/21/99 Julie A. Katz Name of Person Signing Date Total number of pages including cover sheet, attachments, and

REEL: 001979 FRAME: 0505

Tab settings □ □ □ ▼

00-23-1997

SHEET	U.S. DEPARTMENT ( Patent and		
A 12 3	482 Patent and		$\mathcal{D}$
▼	▼	▼	/

To the Honorable Commissioner of Patents and Trauemann.	ttached original documents or copy thereof.
Name of conveying party(ies):    Deliver   Company   Company	Name and address of receiving party(ies)
Bally Entertainment Corporation	Name: Hilton Hotels Corporation
MAY. 1. 2 1997.	Internal Address:
☐ Individual(s) ☐ Assiziation ☐ General Partnership ☐ Limited Annership	Street Address: 9336 Civic Center Drive
☐ Corporation-State Delaware ☐ Other	City: Beverly Hills State: CA ZIP: 90210
Additional name(s) of conveying party(ies) attached? ☐ Yes ☐ No	☐ Individual(s) citizenship
Nature of conveyance:	☐ Association☐ General Partnership
or railed or comoganism	☐ Limited Partnership
☐ Assignment ☐ Merger	U Other
☐ Security Agreement ☐ Change of Name ☐ Other	If assignee is not domiciled in the United States, a domestic representative designation is attached:
Execution Date: June 6, 1996	(Designations must be a separate document from assignment): Additional name(s) & address(es) attached? □ Yes □ No
4. Application number(s) or patent number(s):	
A. Trademark Application No.(s)	B. Trademark Registration No.(s)
See Attached	See Attached
Additional numbers a	Ittached 🕱 Yes 🖸 No
5. Name and address of party to whom correspondence concerning document should be mailed:	6. Total number of applications and registrations involved:
Name:Julie A. Katz, Esq.	
Internal Address: Welsh & Katz, Ltd.	7. Total fee (37 CFR 3.41)\$ 1765.00
	🛱 Enclosed
	☐ Authorized to be charged to deposit account
Street Address: 120 S. Fiverside Plaza	
22nd_Floor	8. Deposit account number:
City: Chicago State: IL ZIP: 60606	23-0920
	(Attach duplicate copy of this page if paying by deposit account)
040 TL 05/2:/97 004844 040 TL 05/2:/97 004844	SE THIS SPACE 0 481 40.00 CK 0 482 1,725.00 CK
9. Statement and signature.  To the best of my knowledge and belief, the foregoing informative original document.	mation is true and correct and any attached copy is a true copy of
Julie A. Katz	5/12/97
Name of Person Signing	Signature Date
Total number of pages including	g cover sheet, attachments, and document:

# EXHIBIT A

Trademark	Application Number	Filing Date	Registration Number	Registration Date	CaseNumber
:					
BALLY'S TOTAL FITNESS	006,844	01-Dec-89	1,623,605	20-Nov-90	59493
BALLY TOTAL FITNESS NUTRITIONALS	157,749	29-Aug-96			68053
BALLY TOTAL FITNESS	157,797	29-Aug-96			68052

Trademark	Application Number	Filing Date	Filing Date Registration Number	Registration Date CaseNumber	CaseNumber
BALLY TOTAL FITNESS	162,713	09-Scp-96			68136
BALLY'S LOGO	421,040	04-Aug-93	1,854,914	20-Sep-94	57934
BALLY'S	421,427	04-Aug-93	1,870,035	27-Dec-94	57716

Trademark	Application Number	Filing Date	Registration Number	Registration Date	CaseNumber
BALLY TOTAL FITNESS	638,246	24-Feb-95			61723
BALLY TOTAL FITNESS	640,187	24-Feb-95	1,922,150	06-Aug-96	62315
BALLY'S PROACTIVE SERVICE	640,368	24-Feb-95			61156
B (STYLIZED)	640,369	24-Feb-95	2.030.899	14-Jan-97	62297
BALLY TOTAL FITNESS	655,244	03-Apr-95	1,973,721	14-May-96	62065

Trademark	Application Number	Filing Date	Registration Number	Registration Date CaseNumber	CaseNumber
BALLY	687,446	02-Oct-87	1,552,562	22-Aug-89	59529
BALLY TOTAL FITNESS AND DESIGN	689,601	16-Jun-95	2,020,345	03-Dec-96	63041
BALLY'S TOTAL FITNESS	800,026	01-Dec-89	1,673,918	28-Jan-92	59486

Trademark	Application Number	Filing Date	Registration Number	Registration Date	CaseNumber	
BALLY (SCRIPT)			639,034	25-Dec-56	60066	
BALLY	024,101	13-Jun-94	1,012,119	03-Jun-75	60065	
BALLY	063,579	29-May-90	1,646,697	04-Jun-91	60064	
VERTICAL CLUB	104,800	09-Oct-90	1,666,141	26-Nov-91	59504	
BALLY	155,686	09-Apr-91	1,909,427	01-Aug-95	60062	
BALLY (SCRIPT)	155,687	09-Apr-91			60063	
B (STYLIZED)	162,874	09-Sep-96			68135	
BALLY TOTAL FITNESS	165,335	13-Sep-96			68142	
B (STYLIZED)	169,129	20-Sep-96			68274	
THE VERTICAL CLUB	174,748	10-Jun-91	1,724,335	13-Oct-92	59482	
VERTICAL CLUB AND DESIGN	210,159	04-Oct-91	1,715,953	15-Sep-92	59484	
FAST FIFTIES	219,597	07-Nov-91			60151	
VERTICAL CLUB AND DESIGN	223,362	16-Jul-79	1,151,961	21-Apr-81	59523	
VERTICAL CLUB	262,009	03-Apr-92	1,734,972	24-N <sub>0</sub> v-92	59483	
BALLY CAJUN	265,904	14-Apr-92	1,792,139	07-Sep-93	63875	

Trademark	Application Number	Filing Date	Registration Number	Registration Date	CaseNumber
BALLY MARDI GRAS	265,906	14-Apr-92	1,806,687	23-Nov-93	63876
BALLY MAVERICK	265,907	14-Apr-92	1,816,368	11-Jan-94	63873
BALLY GAMING	282,077	04-Jun-92	1,820,490	08-Feb-94	63874
VERTICAL CLUB AND DESIGN	290,857	06-Jul-92	1,733,050	25-May-93	60547
BALLY	481,961	20-Jan-94			59427
BALLY'S	482,460	20-Jan-94	1,904,615	l I-Jul-95	59428
BALLY'S (STYLIZED)	485,541	02-Feb-94	1,915,383	29-Aug-95	59549
BALLY'S BELLE OF ORLEANS	487,140	07-Feb-94	2,012.257	29-Oct-96	59450
,					
BALLY	532,200	23-Aug-47	505,218	28-Dec-48	60061

Trademark	Application Number	Filing Date	Registration Number	Registration Date	CaseNumber
VC	608,793	09-Dec-94			61279
7 HEAVEN	619,697		1,440,235	19-May-87	60186
7 HEAVEN DOLLARS	620,651	18-Sep-86	1,439,550	12-May-87	60187
777 HEAVEN	625,038	10-Oct-86	1,439,557	12-May-87	60706
BALLY'S	666,811	12-Jun-87	1,515,843	06-Dec-88	60110
BALLY'S	666,822	15-Jun-87	1,523,910	07-Feb-89	60109
BALLY SYSTEM 5000	693,699	05-Nov-87	1,547,119	11-Jul-89	63877
BALLY'S GRAND	697,657	25-Nov-87	1,533,712	04-Apr-89	60113
BALLY'S GRAND	697,660	25-Nov-87	1,543,909	13-Jun-89	60114
PIN POOL	699,197	01-Dec-55	637,106	13-Nov-56	60178
BALLY'S LAKEHOUSE	723,086	31-Aug-95			63414
BALLY'S	809,446	27-Jun-89	1,581,959	06-Fcb-90	60108
BALLY MIDWAY	M 54 332/9 W	13-Feb-84	1,066,701	06-Aug-84	60728

Trademark	Application Number	Filing Date	Registration Number	Registration Date	CaseNumber
BIG BUCKS	227,176	04-Dec-91	1,847,026	26-Jul-94	60580
BALLY BOB	535,501	09-Jun-94	1,955,688	13-Feb-96	60673
BILLY'S FOOD & BEVERAGE AND CA	536,223	10-Jun-94			60674
BALLY'S SALOON & GAMBLING HALL	538,625	16-Jun-94	1,955,699	13-Feb-96	60675
BILLY'S FOOD & BEVERAGE AND CA	541,524	10-Jun-94			60925
BALLY BOB DESIGN (I)	543,447	28-Jun-94	1,975,755	28-May-96	61015
BALLY'S CASINO LAKESHORE RESOR	655,246	03-Apr-95	1,973,722	14-May-96	62283
BALLY'S CASINO LAKESHORE RESOR	659,487	03-Apr-95	1,982,255	25-Jun-96	62298
BALLY'S COMMODORE CLUB	666,975	27-Apr-95	2,032,928	21-Jan-97	61739
BALLY'S COMMODORE CLUB AND DESIGN	689,578	16-Jun-95	2.034,472	28-Jan-97	63042
BALLY'S SALOON, GAMBLING HALL HOTEL	711,619	07-Aug-95			63277

62910			18-Jul-95	703,332	PARIS CASINO RESORT
63160			18-Jul-95	703,331	PARIS CASINO - RESORT
62556			29-Mar-95	653,425	PARIS
68021			04-Oct-96	208,253	BALLY'S GAMING U.
68024			17-Oct-96	183,087	BALLY'S GAMING U. AND DESIGN 2
68023			04-Oct-96	177,000	BALLY'S GAMING UNIVERSITY
68022			04-Oct-96	176,918	BALLY'S GAMING U. AND DESIGN
66953			14-May-96	103,801	BALLY'S PARIS
n Date CaseNumber	Registration Date	Registration Number	Filing Date	Application Number	Trademark

## AGREEMENT AND PLAN OF MERGER

between

# HILTON HOTELS CORPORATION

and

# **BALLY ENTERTAINMENT CORPORATION**

Dated as of June 6, 1996, and amended as of August 15, 1996

COMPOSITE AGREEMENT

### TABLE OF CONTENTS

Section		Page
	ARTICLE I	
	DEFINITIONS	
1.1 1.2 1.3	Definitions Other Terms Other Definitional Provisions	A-I A-5 A-5
	ARTICLE II	
	THE MERGER	
2.1 2.2 2.3 2.4 2.5 2.6 2.7	Merger Closing Effective Time Effects of the Merger Certificate of Incorporation and By-laws Directors Officers	A-6 A-6 A-6 A-6 A-6 A-6
	ARTICLE III	
	EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT CORPORATIONS; EXCHANGE OF CERTIFICATES	
3.1 3.2	Effect on Capital Stock	А-б А-9
	ARTICLE IV	
	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	
4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13 4.14 4.15 4.16 4.17	Organization, Standing and Corporate Power Subsidiaries Capitalization Authority; Enforceability; No Conflicts; and Consents Vote Required; State Takeover Statutes; Rights Agreement Compliance with Applicable Laws Company SEC Documents; Undisclosed Liabilities Absence of Changes or Events Litigation Taxes Employee Benefits Title to Properties Insurance Brokers and Intermediaries Opinion of Financial Advisor Company Rights Agreement Transactions With Affiliates	A-11 A-11 A-12 A-13 A-13 A-15 A-15 A-15 A-17 A-18 A-18 A-18 A-18
	A-i	

Section		Page
	ARTICLE V	
	REPRESENTATIONS AND WARRANTIES OF ACQUIROR	
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11	Organization, Standing and Corporate Power of Acquiror Capitalization Authority; Enforceability; No Conflicts The Vote Required; Ownership of Company Capital Stock Acquiror SEC Documents; Undisclosed Liabilities Absence of Changes or Events Litigation Taxes Compliance with Applicable Laws ERISA Compliance Brokers	A-19 A-20 A-21 A-21 A-21 A-22 A-22 A-22 A-22
	ARTICLE VI	
	COVENANTS RELATING TO CONDUCT OF BUSINESS	
6.1 6.2 6.3	Conduct of Business of the Company  Conduct of Business of Acquiror  Access to Information	A-23 A-25 A-20
	ARTICLE VII	
	ADDITIONAL AGREEMENTS	
7.1 7.2 7.3 7.4 7.5 7.6 7.7 7.8 7.9 7.10 7.11 7.12 7.13 7.14 7.15 7.16	Preparation of Form S-4 and the Proxy Statement/Prospectus; Stockholders' Meeting Letter of the Company's Accountants Letter of Acquiror's Accountants Reasonable Best Efforts; Notification Approval of Gaming Commissions; Regulatory Matters Supplemental Disclosure Announcements No Solicitation Indemnification; Directors' and Officers' Insurance NYSE Listing Affiliates Rights Agreement Employee Benefits Tax Treatment Transfer Taxes Subsidiary Preferred Stock	A-26 A-27 A-28 A-28 A-29 A-30 A-31 A-31 A-31 A-31 A-31 A-31
	ARTICLE VIII	
	CONDITIONS PRECEDENT	
8.1 8.2 8.3	Conditions to Each Party's Obligation to Effect the Merger  Conditions of Obligations of Acquiror  Conditions of Obligation of the Company	A-32 A-32 A-31

A-ii

Section		Page
	ARTICLE IX	
	TERMINATION AND AMENDMENT	
9.1	Termination	A-34
9.2	Effect of Termination	A-35
9.3	Fees and Expenses	A-35
9.4	Amendment	A-35
9.5	Extension; Waiver	A-36
	ARTICLE X	
	GENERAL PROVISIONS	
10.1	Effectiveness of Representations, Warranties and Agreements	A-36
10.2	Expenses	A-36
10.3	Governing Law	A-36
10.4	Notices	A-36
10.5	Entire Agreement	A-37
10.6	Disclosure Schedule	A-37
10.7	Headings; References	A-37
10.8	Counterparts	A-37
10.9	Parties in Interest; Assignment	A-37
10.10	Severability: Enforcement	A-37
10.11	Specific Performance	A-38
EXHIBIT.	A—Form of Opinion of Latham & Watkins	
EXHIBIT	B—Form of Cpinion of Weil, Gotshal & Manges LLP	
EXHIBIT	C—Form of Tax Certificate of the Company	
EXHIBIT :	D—Form of Tax Certificate of Acquiror	
EXHIBIT :	E-Form of Tax Certificate of Certain Stockholders of the Company	
EVHIDIT	E. Form of Afflicts Letter	

#### AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER, dated as of June 6, 1996, among HILTON HOTELS CORPORATION, a Delaware corporation ("Acquiror"), and BALLY ENTERTAINMENT CORPORATION, a Delaware corporation (the "Company").

#### WITNESSETH:

WHEREAS, upon the terms and subject to the conditions set forth in this Agreement, the Company and Acquiror will enter into a business combination transaction pursuant to which the Company will merge with and into Acquiror (the "Merger"), with Acquiror continuing as the surviving corporation (the "Surviving Corporation");

WHEREAS, the respective Boards of Directors of Acquiror and the Company have determined that the Merger would be fair to and in the best interests of their respective stockholders, and such Boards of Directors have approved this Agreement and the transactions contemplated hereby;

WHEREAS, for federal income tax purposes, it is intended that the Merger shall qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended, and the rules and regulations promulgated thereunder (the "Code"); and

WHEREAS, Acquiror and the Company desire to make certain representations, warranties, covenants and agreements in connection with the Merger and also to prescribe various conditions to the Merger.

NOW, THEREFORE, in consideration of the foregoing and the representations, warranties, covenants and agreements herein contained, the parties hereto agree as follows:

#### ARTICLE I

#### **DEFINITIONS**

- 1.1 Definitions. For purposes of this Agreement, the following terms shall have the meanings set forth below:
  - "Acquiror Disclosure Schedule" shall have the meaning set forth in Section 10.6.
  - "Acquiror SEC Documents" shall have the meaning set forth in Section 5.5.
  - "Acquiror Common Stock" shall have the meaning set forth in Section 3.1(c).
  - "Acquiror Preferred Stock" shall have the meaning set forth in Section 5.2.
  - "Acquisition Proposal" shall have the meaning set forth in Section 7.8(c).
  - "Acquiror Rights" shall have the meaning set forth in Section 3.1(c).
  - "Acquiror Rights Agreement" shall have the meaning set forth in Section 3.1(c).
  - "Acquiror Significant Subsidiaries" shall mean any Subsidiary of Acquiror that constitutes a "Significant Subsidiary" of Acquiror within the meaning of Regulation S-X of the SEC.
    - "Acquiror Stock Options" shall have the meaning set forth in Section 5.2.
    - "Acquiror Stockholder Approval" shall have the meaning set forth in Section 5.3(a).
    - "Acquiror Stockholders' Meeting" shall have the meaning set forth in Section 7.1(e).
  - "Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such other Person.

A-1

- "Applicable Laws" shall mean, with respect to any Person, all statutes, laws, ordinances, rules, orders and regulations of any Governmental Authority applicable to such Person and its business, properties and assets.
- "Business Day" shall mean a day other than a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required by law to close.
  - "Cash Consideration" shall have the meaning set forth in Section 3.1(c).
  - "Certificate of Merger" shall have the meaning set forth in Section 2.3.
  - "Certificates" shall have the meaning set forth in Section 3.2(b).
  - "Closing" shall have the meaning set forth in Section 2.2.
  - "Closing Date" shall have the meaning set forth in Section 2.2.
  - "Code" shall have the meaning set forth in the fourth recital of this Agreement.
- "Common Conversion Number" shall equal 0.25, as may be adjusted pursuant to Sections 3.1(c) and 3.1(d).
- "Company Capital Stock" shall mean the Company Common Stock, the Series D Preferred and the PRIDES.
  - "Company Common Stock" shall have the meaning set forth in Section 3.1(c).
  - "Company Disclosure Schedule" shall have the meaning set forth in Section 10.6.
  - "Company Preferred Stock" shall have the meaning set forth in Section 4.3.
  - "Company Representatives" shall have the meaning set forth in Section 7.8.
  - "Company Rights" shall have the meaning set forth in Section 4.3.
  - "Company SEC Documents" shall have the meaning set forth in Section 4.7.
- "Company Significant Subsidiary" shall mean any Subsidiary of the Company that constitutes a "Significant Subsidiary" of the Company within the meaning of Regulation S-X of the SEC.
  - "Company Stockholder Approval" shall have the meaning set forth in Section 4.4(a).
  - "Company Stockholders' Meeting" shall have the meaning set forth in Section 7.1(d).
  - "Confidentiality Agreement" shall have the meaning set forth in Section 6.3(c).
- "Contamination" shall mean the introduction into the environment (including the land, surface water, ground water, underlying or in proximity to any Real Property and the ambient air above or in the proximity to any Real Property) of any contaminant, pollutant or other toxic or hazardous substance or waste as those terms are defined in applicable Environmental Laws (whether or not upon the Real Property or other property used by the Company or any of its Subsidiaries and whether or not such pollution, when it occurred, violated any Environmental Law) as a result of any actual or threatened spill, discharge, leak, emission, escape, injection, dumping or release of any kind of any substance, in violation of any Environmental Law, or as a result of which the Company or any of its Subsidiaries has or is reasonably likely to become liable to any person or by reason of which the Real Property or any other assets of the Company or any of its Subsidiaries is reasonably likely to suffer or be subjected to any Encumbrance or claim.
  - "Convertible Debentures" shall have the meaning set forth in Section 4.3.
- "Determination Price" shall mean the average of the daily high and low sales prices of the Acquiror Common Stock (regular way) as shown on the New York Stock Exchange Composite Tape (as reported by The Wall Street Journal or, if not reported thereby, any other authoritative source) for the ten consecutive

Trading Days ending on the third Trading Day prior to the Closing Date, as such Determination Price may be adjusted pursuant to Section 3.1(d).

- "DGCL" shall mean the Delaware General Corporation Law.
- "DOJ" shall mean the Department of Justice.
- "Effective Time" shall have the meaning set forth in Section 2.3.
- "Employee Benefit Plans" shall have the meaning set forth in Section 4.11(a).
- "Encumbrances" shall mean any and all mortgages, security interests, liens, claims, pledges, restrictions, leases, title exceptions, rights of others, charges or other encumbrances.
  - "Environmental Laws" shall have the meaning set forth in Section 4.6(c).
- "ERISA" shall mean the Employee Retirement Income Security Act of 1974, and the applicable regulations promulgated thereunder.
- "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
  - "Exchange Agent" shall have the meaning set forth in Section 3.2(a).
  - "Exchange Fund" shall have the meaning set forth in Section 3.2(a).
  - "Form S-4" shall have the meaning set forth in Section 5.3(c).
  - "FTC" shall mean the Federal Trade Commission.
- "GAAP" shall mean generally accepted accounting principles in effect in the United States of America as of the date of the applicable determination.
- "Gaming Commissions" shall mean, collectively, the Nevada State Control Board, the Nevada Gaming Commission, the New Jersey Casino Control Commission, the Mississippi Gaming Commission and the Louisiana Riverboat Gaming Commission.
- "Gaming Laws" shall mean any Federal, state, local or foreign statute, ordinance, rule, regulation, policy, permit, consent, approval license, judgment, order, decree, injunction or other authorization governing or relating to the casino and gaming and racetrack activities and operations of the Company or Acquiror, as applicable, including (i) the New Jersey Casino Control Act and the rules and regulations promulgated thereunder (the "New Jersey Gaming Laws"), (ii) the Nevada Gaming Control Act and the rules and regulations promulgated thereunder, and the Clark County, Nevada Code and the rules and regulations promulgated thereunder (the "Nevada Gaming Laws"), (iii) the Mississippi Gaming Control Act and the rules and regulations promulgated thereunder (the "Mississippi Gaming Laws"), and (iv) the Louisiana Riverboat Economic Development and Gaming Act and the rules and regulations promulgated thereunder (the "Louisiana Gaming Laws").
  - "Gains Tax" shall have the meaning set forth in Section 4.4(c).
- "Governmental Authority" shall mean any foreign, Federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality.
  - "Hazardous Material" shall have the meaning set forth in Section 4.6(d).
  - "HSR Act" shall have the meaning set forth in Section 4.4(c).
- "Improvements" shall mean, with respect to any Real Property, all buildings, fixtures, improvements and facilities located on or attached to such Real Property or owned or leased by the Company or any of its Subsidiaries and used in, on or at such Real Property, together with any and all loading docks, parking lots, garages, and other facilities serving any such buildings; and landscaping and site improvements.

- "Indemnified Liabilities" shall have the meaning set forth in Section 7.9(a).
- "Indemnified Parties" shall have the meaning set forth in Section 7.9(a).
- "IRS" means the United States Internal Revenue Service.
- "Legal Proceedings" means any judicial, administrative or arbitral actions, suits, proceedings (public or private) or governmental proceedings.
- "Material Adverse Effect" shall mean, with respect to any Person, any change, occurrence or effect that is or is reasonably likely to be materially adverse to the assets, business, results of operations or condition (financial or otherwise) of such party and its Subsidiaries taken as a whole.
  - "Merger" shall have the meaning set forth in the first recital to this Agreement.
  - "Merger Consideration" shall have the meaning set forth in Section 3.2(b).
- "New PRIDES" shall mean Acquiror's Preferred Redeemable Increased Dividend Equity Securities, 8% PRIDES, Convertible Preferred Stock; the rights, preferences and terms of which shall be the same as those of the PRIDES, except that the issuer thereof shall be Acquiror, the vote per share of New PRIDES shall be the product of 0.80 and the Common Conversion Number, and the conversion and redemption terms shall reflect the adjustments provided for in Section 3(e) of the Certificate of Designations, Preferences, Rights and Limitations of PRIDES.
  - "NYSE" shall mean the New York Stock Exchange, Inc.
  - "Option" shall have the meaning set forth in Section 3.1(f).
  - "Permits" shall have the meaning set forth in Section 4.6(a).
- "Permitted Encumbrances" shall mean only the following title exceptions: (a) taxes either not delinquent or being diligently contested; (b) mechanics, materialmen's or similar statutory liens being diligently contested; (c) other exceptions that do not and would not, individually or in the aggregate, have a Material Adverse Effect with respect to the Company; and (d) Encumbrances related to indebtedness disclosed in the Company SEC Documents filed and publicly available prior to the date of this Agreement.
- "Person" shall mean an individual, corporation, partnership, trust or unincorporated organization or a government or any agency or political subdivision thereof.
- "PRIDES" shall mean the Company's Preferred Redeemable Increased Dividend Equity Securities, 8% PRIDES, Convertible Preferred Stock.
- "Proxy Statement/Prospectus" shall mean the joint proxy statement relating to the Company Stockholder Approval and Acquiror Stockholder Approval and the prospectus relating to the issuance of shares of the Acquiror Common Stock and New PRIDES in connection with the consummation of the transactions contemplated by this Agreement, as such joint proxy statement and prospectus may be amended or supplemented from time to time.
  - "Real Property" shall have the meaning set forth in Section 4.12.
  - "SAR" shall have the meaning set forth in Section 3.1(f).
- "Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.
  - "SEC" shall mean the Securities and Exchange Commission.
  - "Series D Preferred" shall mean the Company's Series D Convertible Exchangeable Preferred Stock.
  - "Stock Consideration" shall have the meaning set forth in Section 3.1(c).
- "Subsidiary" shall mean, with respect to any Person, (i) each corporation, partnership, joint venture, limited liability company or other legal entity of which such Person owns, either directly or indirectly, 50%

or more of the stock or other equity interests the holders of which are generally entitled to vote for the election of the board of directors or similar governing body of such corporation, partnership, joint venture or other legal entity and (ii) each partnership or limited liability company in which such Person or another Subsidiary of such Person is the general partner, managing partner or other otherwise controls (including, without limitation, Belle of Orleans, LLC and Bally's Olympia Limited Partnership).

- "Subsidiary Preferred Stock" shall have the meaning set forth in Section 4.3.
- "Surviving Corporation" shall have the meaning set forth in the first recital of this Agreement.
- "Tax" or "Taxes" shall mean all taxes, charges, fees, imposts, levies, gaming or other assessments, including, without limitation, all net income, gross receipts, capital, sales, use, ad valorem, value added, transfer, franchise, profits, inventory, capital stock, license, withholding, payroll, employment, social security, unemployment, excise, severance, stamp, occupation, property and estimated taxes, customs duties, fees, assessments and charges of any kind whatsoever, together with any interest and any penalties, fines, additions to tax or additional amounts imposed by any taxing authority (domestic or foreign) and shall include any transferee liability in respect of Taxes, any liability in respect of Taxes imposed by contract, tax sharing agreement, tax indemnity agreement or any similar agreement.
- "Tax Return" shall mean any report, return, document, declaration or any other information or filing required to be supplied to any taxing authority or jurisdiction (foreign or domestic) with respect to Taxes, including without limitation, information returns, any document with respect to or accompanying payments or estimated Taxes, or with respect to or accompanying requests for the extension of time in which to file any such report, return document, declaration or other information.
  - "Termination Date" shall have the meaning set forth in Section 9.1(d).
  - "Termination Notice" shall have the meaning set forth in Section 3.1(d).
  - "Third Party" shall mean a party or parties unaffiliated with either the Company or Acquiror.
  - "Top-Up Intent Notice" shall have the meaning set forth in Section 3.1(c).
  - "Trading Day" shall mean a day on which the NYSE is open for the transaction of business.
- 1.2 Other Terms. Other terms may be defined elsewhere in the text of this Agreement and, unless otherwise indicated, shall have such meaning throughout this Agreement.
- 1.3 Other Definitional Provisions. (a) The words "hereof," "herein," and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement.
- (b) The terms defined in the singular shall have a comparable meaning when used in the plural, and vice versa.
  - (c) The terms "dollars" and "S" shall mean United States dollars.

#### ARTICLE II

#### THE MERGER

- 2.1 Merger. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the DGCL, the Company shall be merged with and into Acquiror at the Effective Time. Following the Merger, the separate corporate existence of the Company shall cease and Acquiror shall continue as the Surviving Corporation and shall succeed to and assume all the rights and obligations of the Company in accordance with the DGCL.
- 2.2 Closing. Unless this Agreement shall have been terminated and the transactions herein contemplated shall have been abandoned pursuant to Section 9.1, the closing of the Merger (the "Closing") will take place at 10:00 a.m., New York City time, on the third Business Day following the date on which the last of the conditions set forth in Article VIII is fulfilled or waived (the "Closing Date"), at the offices of Weil, Gotshal & Manges, 767 Fifth Avenue, New York, New York 10153, unless another date, time or place is agreed to by the parties hereto.
- 2.3 Effective Time. On the Closing Date, or as soon as practicable thereafter, the parties hereto shall cause the Merger to be consummated by filing a certificate of merger (the "Certificate of Merger") executed in accordance with the relevant provisions of the DGCL with the Secretary of State of the State of Delaware. The Merger shall become effective at such time as the Certificate of Merger is so duly filed, or at such time thereafter as is provided in the Certificate of Merger (the "Effective Time").
  - 2.4 Effects of the Merger. The Merger shall have the effects as set forth in Section 259 of the DGCL.
- 2.5 Certificate of Incorporation and By-laws. (a) The Certificate of Incorporation of Acquiror, as in effect immediately prior to the Effective Time, shall be the Certificate of Incorporation of the Surviving Corporation after the Effective Time, until duly amended in accordance with its terms and the DGCL.
- (b) The By-laws of Acquiror, as in effect immediately prior to the Effective Time, shall be the By-laws of the Surviving Corporation, until thereafter amended as provided therein, by Applicable Law or the Certificate of Incorporation of the Surviving Corporation.
- 2.6 Directors. The directors of Acquiror immediately prior to the Effective Time shall be the directors of the Surviving Corporation, until the earlier of their resignations or removal or until their respective successors are duly elected and qual fied, as the case may be. Immediately after the Effective Time, Acquiror shall take all action necessary to elect one individual designated by the Board of Directors of the Company and reasonably acceptable to Acquiror, as director of the Surviving Corporation.
- 2.7 Officers. The officers of Acquiror immediately prior to the Effective Time shall be the officers of the Surviving Corporation, until the earlier of their resignation or removal or until their respective successors are duly elected and qualified, as the case may be.

#### ARTICLE III

# EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT CORPORATIONS; EXCHANGE OF CERTIFICATES

- 3.1 Effect on Capital Stock. At the Effective Time, by virtue of the Merger and without any action on the part of the holder of any shares of Company Capital Stock or the holder of any shares of the capital stock of Acquiror:
  - (a) Capital Stock of Acquiror. Each share of the capital stock of Acquiror issued and outstanding immediately prior to the Effective Time shall remain an issued and outstanding share of the same class of capital stock of the Surviving Corporation.

A-6

- (b) Cancellation of Treasury Stock and Acquiror-Owned Stock. Each share of Company Capital Stock that is directly owned by the Company and each share of Company Capital Stock that is directly owned by Acquiror or any subsidiary of Acquiror shall be canceled and retired and shall cease to exist and no consideration shall be delivered or deliverable in exchange therefor.
- (c) Conversion of Company Common Stock. (i) Subject to Section 3.3, each share of Common Stock, par value \$.66% per share ("Company Common Stock"), of the Company, issued and outstanding immediately prior to the Effective Time (excluding shares cancelled in accordance with Section 3.1(b)) together with the Company Rights attached thereto or associated therewith shall be converted into the right to receive a number of fully paid and non-assessable shares of Common Stock, par value \$2.50 per share, of Acquiror ("Acquiror Common Stock") equal to the Common Conversion Number (the "Stock Consideration'); provided, however, that in the event the Determination Price is less than \$108 (as adjusted pursuant to Section 3.1(d), the "Target Amount", each share of Company Common Stock shall be converted into the right to receive a number of fully paid and non-assessable shares of Acquiror Common Stock equal to the Common Conversion Number plus an amount in cash, not to exceed \$3.00, equal to (i) the difference between the Target Amount and the Determination Price multiplied by (ii) the Common Conversion Number (rounded to the nearest hundredth, or if there shall not be a nearest hundredth, to the next highest hundredth but in no event greater than \$3.00) (the "Cash Consideration"); and, provided, further, that in the event the Determination Price is less than \$80 (as adjusted pursuant to Section 3.1(d), the "Floor Amount", Acquiror shall have the right to give written notice to the Company (the "Top-Up Intent Notice") that the board of directors of Acquiror elects to increase the Common Conversion Number such that the product of the Common Conversion Number and the Determination Price shall equal \$20 (it being understood that notwithstanding such increase in the Common Conversion Number, the Cash Consideration shall remain the same). The Top-Up Intent Notice shall be delivered to the Company no later than 2:00 p.m. on the second Business Day prior to the Closing Date. If, in such case, Acquiror does not deliver a Top-Up Intent Notice, the Company shall have the right to give written notice to Acquiror (the "Termination Notice") that the Company elects to terminate this Agreement. The Termination Notice shall be delivered to Acquiror no later than 2:00 p.m. on the Business Day prior to the Closing Date.
- (ii) Pursuant to the Rights Agreement, dated as of July 14, 1988 (the "Acquiror Rights Agreement"), between Acquiror and the First National Bank of Chicago, as rights agent, one right issued under the Acquiror Rights Agreement (an "Acquiror Right") will be attached to each share of Acquiror Common Stock issued upon conversion of Company Common Stock in accordance with Section 3.1(c) and all references in this Agreement to Acquiror Common Stock shall be deemed to include the Acquiror Rights. As of the Effective Time, all shares of Company Common Stock and all Company Rights converted pursuant to Section 3.1(c) shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist, and each holder of a certificate representing any such shares of Company Common Stock and any such Company Rights shall cease to have any rights with respect thereto, except the right to receive, upon the surrender of any such certificates, certificates representing the shares of Acquiror Common Stock and any cash in lieu of fractional shares of Acquiror Common Stock to be issued or paid in consideration therefor upon surrender of such certificate in accordance with Section 3.2(e), and any dividends or other distributions to which such holder is entitled pursuant to Section 3.2(c), in each case without interest.
- (d) Certain Adjustments and Determinations. If, between the date of this Agreement and the Effective Time, the outstanding shares of Acquiror Common Stock or Company Common Stock shall have been changed into a different number of shares or a different class, by reason of any stock dividend, subdivision, reclassification, recapitalization, split, combination or exchange of shares, the Common Conversion Number, the Determination Price, the Target Amount and the Floor Amount specified in Section 3.1(c) correspondingly shall be adjusted to reflect such stock dividend, subdivision, reclassification, recapitalization, split, combination or exchange of shares. Acquiror intends to effect a 4-for-1 stock split of Acquiror Common Stock, and upon such stock split, the Common Conversion Number shall be 1.0 (as may

be further adjusted pursuant to Section 3.1(c)), the Target Amount shall be \$27 and the Floor Amount shall be \$20; and, in the event the Determination Price is calculated on the basis of pre-split trading prices, the Determination Price shall be such amount divided by four or, if the Determination Price is calculated on the basis of pre-split and post-split trading prices, appropriate adjustments shall be made to calculate a comparable number.

1

- (e) Conversion of PRIDES. Each share PRIDES issued and outstanding at the Effective Time (other than shares to be canceled in accordance with Section 3.1(b)) shall be converted into the right to receive one validly issued, fully paid and nonassessable share of New PRIDES, subsequent to the Effective Time. As of the Effective Time, all such shares of PRIDES shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist, and each holder of a certificate representing any such shares of PRIDES shall cease to have any rights with respect thereto, except the right to receive the shares of New PRIDES, to be issued in consideration thereof upon surrender of such certificate in accordance with Section 3.2(a) and any dividends or other distributions to which such holder is entitled to pursuant to Section 3.2(c), in each case without interest.
- (f) Stock Options; SARS. At the Effective Time, each then outstanding option to purchase shares of Company Common Stock under the Company's 1989 Incentive Plan, 1993 Non-Employee Directors' Stock Option Plan and 1985 Incentive Plan (collectively, the "Company Stock Option Plans"), whether or not then vested or exercisable in accordance with its terms (collectively, the "Options"), shall become exercisable in full and shall be settled in exchange for an amount of cash equal to the product of (1) the amount by which (A) the sum of (i) the product of the Common Conversion Number and the Determination Price plus (ii) the Cash Consideration (if any) exceeds (B) the exercise price per share of such Option and (2) the number of shares of Company Common Stock issuable pursuant to the unexercised portion of such Option, subject to any required withholding of taxes (such amount being hereinafter referred to as, the "Option Consideration"). At the Effective Time, each then outstanding stock appreciation right granted by the Company under any of the Company Stock Option Plans, whether or not then vested or exercisable in accordance with its terms (collectively, the "SARs"), shall become exercisable in full and shall be settled in exchange for an amount of cash equal to the product of (1) the amount by which (A) the sum of (i) the product of the Common Conversion Number and the Determination Price plus (ii) the Cash Consideration (if any) exceeds (B) the appreciation base per share of Company Common Stock for such SAR and (2) the number of shares of Company Common Stock covered by such SAR, subject to any required withholding of taxes (such amount being hereinafter referred to as, the "SAR Consideration"); provided, however, that with respect to each SAR issued in tandem with an Option, the payment pursuant to the immediately preceding sentence with respect to the corresponding Option shall be deemed to constitute full satisfaction and settlement of such SAR and the holder of such SAR will be entitled to no additional payment with respect to the SAR, and such SAR shall be cancelled. Notwithstanding the foregoing, with respect to any person subject to Section 16(a) of the Exchange Act, any amount of Option Consideration or SAR Consideration, as the case may be, shall be paid as soon as practicable after the first date payment can be made without liability to such person under Section 16(b) of the Exchange Act. From and after the Effective Time, the Options and SARs shall represent only the right of the holders of such Options and SARs to receive the payments specified above upon the surrender thereof. Upon receipt of the Option Consideration or the SAR Consideration (if any), as the case may be, the corresponding Option or the corresponding SAR, as the case may be, shall be canceled. The surrender of an Option or a SAR, as the case may be, to the Company in exchange for the Option Consideration or the SAR Consideration, as the case may be, shall be deemed a release of any and all rights the holder had or may have had in respect of such Option or such SAR. Except as otherwise agreed to by the parties hereto, (i) all Company Stock Option Plans shall terminate as of the Effective Time and the provisions in any other plan, program or arrangement providing for the issuance or grant of any other interest in respect of the capital stock of the Company or, except as set forth in Section 3.1(f) of the Company Disclosure Schedule, any Subsidiary thereof, shall be canceled as of the Effective Time, and (ii) the Company shall take all permitted action necessary to ensure that following the Effective Time no participant in any Company Stock Option Plan or other plans, programs or

arrangements shall have any right thereunder to acquire equity securities of the Company, the Surviving Corporation or any Subsidiary thereof and to terminate all such plans.

- 3.2 Exchange of Certificates. (a) Exchange Agent. Prior to the Effective Time, Acquiror shall designate a bank or trust company to act as exchange agent in the Merger (the "Exchange Agent"), and Acquiror shall deposit with the Exchange Agent as of the Effective Time (or otherwise when requested by the Exchange Agent from time to time in order to effect any exchange pursuant to this Section 3.2) for the benefit of the holders of shares of Company Common Stock and PRIDES for exchange in accordance with this Article III, through the Exchange Agent, certificates evidencing the shares of Acquiror Common Stock and New PRIDES issuable and the Cash Consideration (if any) deliverable pursuant to Sections 3.1(c) and 3.1(d) in exchange for outstanding shares of Company Common Stock and PRIDES. Such shares of Acquiror Common Stock and PRIDES and Cash Consideration (if any), together with any dividends or distributions with respect thereto with a record date after the Effective Time, shall hereinafter be referred to as the "Exchange Fund." The Exchange Agent shall, pursuant to irrevocable instructions, deliver the shares of Acquiror Common Stock and New PRIDES issuable and the Cash Consideration (if any) deliverable pursuant to Sections 3.1(c) and 3.1(d) out of the Exchange Fund.
- (b) Exchange Procedures. As soon as reasonably practicable after the Effective Time, Acquiror shall instruct the Exchange Agent to mail to each holder of record of a certificate or certificates that immediately prior to the Effective Time evidenced outstanding shares of Company Capital Stock (the "Certificates") whose shares were converted into the right to receive shares of Acquiror Common Stock and Cash Consideration (if any) or New PRIDES, as the case may be, pursuant to Section 3.1, (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon proper delivery of the Certificates to the Exchange Agent and shall be in such form and have such other provisions as Acquiror reasonably may specify) and (ii) instructions for use in effecting the surrender of the Certificates in exchange for certificates evidencing shares of Acquiror Common Stock or New PRIDES, as the case may be, and payment of Cash Consideration, if applicable. Upon surrender of a Certificate for cancellation to the Exchange Agent, together with such letter of transmittal, duly executed, and such other customary documents as may be required by the Exchange Agent, the holder of such Certificate shall be entitled to receive in exchange therefor (A) a certificate evidencing that number of whole shares of Acquiror Common Stock or New PRIDES, as applicable, that such holder has the right to receive pursuant to the provisions of Section 3.1, (B) in the case of Acquiror Common Stock, payment evidencing the Cash Consideration (if any) and cash in lieu of fractional shares of Acquiror Common Stock to which such holder is entitled pursuant to Sections 3.1(c) and 3.2(e), and (C) any dividends or other distributions to which such holder is entitled pursuant to Section 3.2(c) (the shares of Acquiror Common Stock or New PRIDES, as the case may be, cash, dividends and distributions described in clauses (A), (B) and (C) being hereinafter collective y, the "Merger Consideration", and the Certificates so surrendered shall forthwith be canceled. In the event of a transfer of ownership of shares of Company Capital Stock that is not registered in the transfer records of the Company, certificates evidencing the proper number of shares of Acquiror Common Stock or New PRIDES, as applicable, may be issued in accordance with this Article III to a person other than the person in whose name the Certificate so surrendered is registered, if the Certificate is presented to the Exchange Agent, accompanied by all documents required to evidence and effect such transfer and by evidence that any applicable stock transfer taxes have been paid or are not applicable. Until surrendered as contemplated by this Section 3.2(b), each Certificate shall be deemed at any time after the Effective Time to evidence only the right to receive upor such surrender the appropriate Merger Consideration.
- (c) Distributions with Respect to Unexchanged Shares. No dividends or other distributions declared or made after the Effective Time with respect to Acquiror Common Stock with a record date after the Effective Time shall be paid to the holder of any unsurrendered Certificate with respect to the shares of Acquiror Common Stock or New PRIDES evidenced thereby, and no other part of the Merger Consideration shall be paid to any such holder, until the holder of such Certificate shall surrender such Certificate in accordance with this Article III. Subject to the effect of Applicable Laws, following surrender of any such Certificate, there shall be paid to the holder of the certificates evidencing whole shares of Acquiror Common Stock or New PRIDES, as applicable, issued in exchange therefor, without interest, (i) at the time of such surrender, the amount of dividends or other

distributions with both a record date and payment date after the Effective Time, but prior to surrender, payable with respect to such whole shares of Acquiror Common Stock or New PRIDES, as applicable, and (ii) at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to surrender and a payment date occurring after surrender, payable with respect to such whole shares of Acquiror Common Stock or New PRIDES as applicable. No interest shall be paid on the Merger Consideration or such dividends or other distributions.

- (d) No Further Rights in Company Capital Stock. All shares of Acquiror Common Stock and New PRIDES issued upon the surrender for exchange of Certificates in accordance with the terms of this Article III (and any cash paid pursuant to Section 3.2(c) or, in the case of Acquiror Common Stock, Sections 3.1(c) and 3.2(e)) shall be deemed to have been issued (or paid) in full satisfaction of all rights pertaining to such shares of Company Capital Stock and Company Rights exchanged therefor represented by such Certificates; subject, however, to the Surviving Corporation's obligation to pay any dividends or make any other distributions with a record date prior to the Effective Time which may have been declared or made by the Company on such shares of Company Capital Stock in accordance with the terms of this Agreement or prior to the date of this Agreement and which remain unpaid at the Effective Time and have not been paid prior to surrender, and there shall be no further registration of transfers on the stock transfer books of the Surviving Corporation of the shares of Company Capital Stock which were outstanding immediately prior to the Effective Time. If, after the Effective Time, Certificates are presented to the Surviving Corporation or the Exchange Agent for any reason, they shall be canceled and exchanged as provided in this Article III, except as otherwise provided by law.
- (e) No Fractional Shares. (i) No certificates or scrip representing fractional shares of Acquiror Common Stock shall be issued upon the surrender for exchange of Certificates, and such fractional share interests will not entitle the holder thereof to vote or to any rights of a shareholder of Acquiror.
- (ii) Notwithstanding any provision of this Agreement, each holder of shares of Company Common Stock exchanged pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Acquiror Common Stock (after taking into account all Certificates delivered by such holder) shall receive, in lieu thereof, cash (without interest) in an amount equal to such fractional part of a share of Acquiror Common Stock multiplied by the Determination Price.
- (f) No Liability. Neither Acquiror nor the Company shall be liable to any holder of shares of Company Capital Stock or Acquiror Common Stock, as the case may be, for such shares of Acquiror Common Stock or New PRIDES (or dividends or distributions with respect thereto) or cash in lieu of fractional shares of Acquiror Common Stock which have been delivered to a public official pursuant to any applicable abandoned property, escheat or similar law.
- 3.3. Dissenting Shares. Notwithstanding any other provisions of this Agreement to the contrary, in the event shares of Company Common Stock are converted into the right to receive both the Stock Consideration and Cash Consideration pursuant to Section 3.1(c), shares of Company Common Stock that are outstanding immediately prior to the Effective Time and that are held by stockholders who shall have not voted in favor of the Merger and who shall have demanded properly in writing appraisal for such shares in accordance with Section 262 of the DGCL (collectively, the "Dissenting Shares") shall not be converted into or represent the right to receive the Stock Consideration and Cash Consideration. Such stockholders instead shall be entitled to receive payment of the appraised value of such shares of Company Common Stock held by them in accordance with the provisions of such Section 262, except that all Dissenting Shares held by stockholders who shall have failed to perfect or who effectively shall have withdrawn or lost their rights to appraisal of such shares of Company Common Stock under such Section 262 shall thereupon be deemed to have been converted into and to have become exchangeable, as of the Effective Time, for the right to receive, without any interest thereon, the Stock Consideration and Cash Consideration upon surrender in the manner provided in this Article III, of the Certificate or Certificates that formerly evidenced such shares of Company Common Stock. The Company shall give Acquiror (i) prompt notice of any demands for appraisal of shares of Company Common Stock received by the Company and (ii the opportunity to direct all negotiations and proceedings with respect to any such demands. The Company shall not, without the prior written consent of Acquiror, make any payment with respect to, or settle, offer to settle, or otherwise negotiate any such demands.

#### ARTICLE IV

#### REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company hereby represents and warrants to Acquiror as follows:

- 4.1 Organization, Standing and Corporate Power. Each of the Company and each Company Significant Subsidiary is a corporation or partnership duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is organized and has the requisite corporate or partnership power and authority to carry on its business as now being conducted. Each of the Company and its Subsidiaries is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the nature of its business or the ownership or leasing of its properties makes such qualifications or licensing necessary, other than in such jurisdictions where the failure to be so qualified or licensed (individually or in the aggregate) would not have a Material Adverse Effect with respect to the Company. The Company has delivered to Acquiror complete and correct copies of its Restated Certificate of Incorporation and By-laws, in each case as amended to the date of this Agreement.
- 4.2 Subsidiaries. Section 4.2 of the Company Disclosure Schedule sets forth all the Subsidiaries of the Company. Except as set forth in Section 4.2 of the Company Disclosure Schedule, or as described in Section 4.3 or as disclosed in the Company SEC Documents, all the outstanding shares of capital stock or other equity interests of each Subsidiary of the Company are duly authorized, validly issued, fully paid and non-assessable and are owned by the Company, by another wholly-owned Subsidiary of the Company or by the Company and another wholly-owned Subsidiary of the Company, free and clear of all Encumbrances. Subject to compliance with applicable Gaming Laws, the respective certificates of incorporation and bylaws or other organizational documents of the Company's Subsidiaries do not contain any provision limiting or otherwise restricting the ability of Acquiror, following the Effective Time, from controlling such Subsidiaries on the same basis as the Company.
- 4.3 Capitalization. The authorized capital stock of the Company consists of 120,000,000 shares of Company Common Stock and 30.000,000 shares of preferred stock, par value \$1.00 per share ("Company Preferred Stock''). As of May 31, 1996, (i) 49,763,359 shares of Company Common Stock were issued and outstanding, (ii) 33,358 shares of Company Common Stock were held by the Company in its treasury, (iii) 496,988 shares of the Series D Preferred, convertible into an aggregate of 1,103,313 shares of Company Common Stock, were issued and outstanding, and 1,103,313 shares of Company Common Stock were reserved for issuance upon conversion of the Series D Preferred, (iv) 15,525,000 shares of PRIDES, currently convertible into an aggregate of 14,283,000 shares of Company Common Stock, were issued and outstanding, and 17,388,000 shares of Company Common Stock were reserved for issuance upon conversion of the PRIDES, (v) 4,985,878 shares of Company Common Stock were reserved for issuance upon exercise of outstanding Options, (vi) no shares of Company Common Stock were reserved for issuance upon exercise of outstanding SARS (other than 508,334 shares granted in tandem with Options), (vii) 59,600 shares of Company Common Stock were reserved for issuance in connection with the Company's Employee Stock Purchase Plans, (viii) 2,618,405 shares of Company Common Stock were reserved for issuance upon conversion of the Company's 6% Convertible Subordinated Debentures due 1998 (the "6% Convertible Debentures") and the Company's 10% Convertible Subordinated Debentures due 2006 (the "10% Convertible Debentures" and, together with the 6% Convertible Debentures, the "Convertible Debentures"), the 6% Convertible Debentures being convertible into an aggregate of 69,119 shares of Company Common Stock, and the 10% Convertible Debentures being convertible into an aggregate of 2,549,286 shares of Company Common Stock, and (ix) 800,000 shares of Company Preferred Stock (Series B Junior Participating) were reserved for issuance in connection with the rights to purchase shares of certain Company Preferred Stock (the "Company Rights") issued pursuant to the Rights Agreement dated as of December 4, 1986, as amended (as amended to the date of this Agreement and as further amended from time to time, the "Rights Agreement"), between the Company and Chemical Bank, as Rights Agent ('he "Company Rights Agent"). Except as set forth above, as of May 31, 1996, no shares of capital stock or other voting or equity securities of the Company were issued, reserved for issuance

A-11

or outstanding. There are no outstanding SARs which were not granted in tandem with (and which would not terminate upon exercise of) a related Option and there are no other outstanding contractual rights the value of which is derived from the financial performance of the Company or the value of shares of Company Common Stock. All outstanding shares of capital stock of the Company are duly authorized, validly issued, fully paid and nonassessable and are not subject to preemptive rights. Except as set forth above, there are no bonds, debentures, notes or other indebtedness of the Company or any of its Subsidiaries having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which stockholders of the Company may vote. Except as set forth above or in Section 4.3 of the Company Disclosure Schedule and other than 1,505,405 shares of Company Common Stock and 376,351 shares of Common Stock of Bally Total Fitness Holding Corporation deliverable upon exchange of Bally Casino, Inc.'s Series A Cumulative Exchangeable Preferred Stock (the "Subsidiary Preferred Stock"), as of the date of this Agreement, there are no outstanding securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind to which the Company or any of its Subsidiaries is a party or by which any of them is bound, obligating the Company or any of its Subsidiaries to issue, deliver or sell or cause to be issued, delivered or sold, additional shares of capital stock or other voting or equity securities of the Company or any of its Subsidiaries or obligating the Company or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking. Except for the redemption of the Series D Preferred, the Company Rights or as set forth above or in Section 4.3 of the Company Disclosure Schedule and other than redemptions, purchases and other acquisitions required by applicable provisions under Gaming Laws or similar provisions contained in the terms of the capital stock of the Company or any of its Subsidiaries, there are not any outstanding contractual obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of the Company or any of its Subsidiaries.

- 4.4 Authority; Enforceability; No Conflicts; and Consents. (a) The Company has the requisite corporate power and authority to enter into this Agreement and, subject to obtaining the Company Stockholder Approval, to consummate the transactions contemplated by this Agreement. The execution and delivery of this Agreement by the Company and the consummation by the Company of the transactions contemplated by this Agreement have been duly authorized by all necessary corporate action on the part of the Company, subject, in the case of the consummation of the Merger, to approval of this Agreement by the holders of a majority of the voting power of the outstanding shares of Company Common Stock and the PRIDES, voting as a class (the "Company Stockholder Approval"), at a special meeting of the holders of Company Common Stock and the PRIDES. This Agreement has been duly executed and delivered by the Company and, assuming this Agreement constitutes the valid and binding obligations of Acquiror, constitutes the valid and binding obligations of the Company, enforceable against the Company in accordance with its terms.
- (b) The execution and delivery of this Agreement do not, and the consummation of the transactions contemplated by this Agreement and compliance with the provisions of this Agreement will not, conflict with, or result in any violation of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cance lation or acceleration of any obligation or cause loss of a material benefit under, or result in the creation or maturation of any lien or purchase right upon any of the properties or assets of the Company or any of its Subsidiaries under, (i) the Restated Certificate of Incorporation or By-laws of the Company or the comparable charter or organizational documents of any of its Subsidiaries, (ii) other than severance agreements, severance plans and employment agreements disclosed in the Company SEC Documents or in the Company Disclosure Schedule or otherwise previously disclosed to Acquiror and bank credit agreements, financing leases and indentures relating to notes and debentures identified under "Long Term Debt" of the Notes to Consolidated Financial Statements of the Company included in its 1995 Annual Report or as set forth in Section 4.4 of the Company Disclosure Schedule and subject to the governmental filings and other matters referred to in Section 4.4(c), any loan or credit agreement, note, bond, mortgage, indenture, lease or other agreement, instrument, permit, concession, franchise or license applicable to the Company or any of its Subsidiaries or their respective properties or assets or (iii) subject to the governmental filings and other matters referred to in Section 4.4(c), any judgment, order, decree, statute, law, ordinance, rule or regulation applicable to the Company or any of its Subsidiaries or their respective properties or assets, other than, in the case of

- clauses (ii) or (iii), any such conflicts, violations, defaults, rights or liens that individually or in the aggregate would not (x) have a Material Adverse Effect with respect to the Company, (y) impair, in any material respect, the ability of the Company to perform its obligations under this Agreement or (z) prevent or significantly delay the consummation of any of the transactions contemplated by this Agreement.
- (c) No consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Authority is required by the Company or any of its Subsidiaries in connection with the execution and delivery of this Agreement by the Company or the consummation by the Company of the transactions contemplated by this Agreement, except for (i) the filing of a premerger notification and report form by the Company under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 (the "HSR Act"), (ii) the filing with the SEC of (x) a proxy statement relating to the Company Stockholder Approval and (y) such reports and filings under Section 13 and 16 of the Exchange Act as may be required in connection with this Agreement and the transactions contemplated hereby, (iii) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware and appropriate documents with the relevant authorities of other states in which the Company. is qualified to do business, (iv) the approval by (A) the New Jersey Casino Control Commission under the New Jersey Gaming Laws, (B) the Nevada State Gaming Control Board and the Nevada Gaming Commission under the Nevada Gaming Laws, (C) the Clark County Liquor and Gaming Licensing Board pursuant to the Clark County, Nevada Code and the rules and regulations promulgated thereunder, (D) the Mississippi Gaming Commission under the Mississippi Gaming Laws, and (E) the Louisiana Riverboat Gaming Commission under the Louisiana Gaming Laws, (v) the filing of notices and the approval of Governmental Authorities as may be required under the applicable Gaming Laws of the States of Maryland and Washington and under the Indian Gaming Regulatory Act of 1988, (vi) as may be required by any applicable state securities or "blue sky" laws, (vii) in connection with any state or local tax which is attributable to the beneficial ownership of real property of the Company or its Subsidiaries ("Clains Tax"), (viii) such immaterial filings and consents as may be required under any environmental, health or safety law or regulation pertaining to any notification, disclosure or required approval triggered by the Merger or the transactions contemplated by this Agreement, (ix) such immaterial filings, consents, approvals, orders, registrations and declarations as may be required under the laws of any foreign country in which the Company or any of its Subsidiaries conducts any business or owns any assets, and (x) such other consents, approvals, orders, authorizations, registrations, declarations and filings the failure of which to be obtained or made would not, individually or in the aggregate, (1) have a Material Adverse Effect with respect to the Company, (2) impair, in any material respect, the ability of the Company to perform its obligations under this Agreement or (3) prevent or significantly delay the consummation of the transactions contemplated by this Agreement.
- 4.5 Vote Required; State Takeover Statutes; Rights Agreement. (a) The Company Stockholder Approval is the only vote of the holders of any class or series of the Company's capital stock necessary to approve this Agreement and the transactions contemplated hereby.
- (b) The Board of Directors of the Company has approved the Merger and this Agreement and such approval is sufficient to render the provisions of Section 203 of the DGCL inapplicable to the Merger, this Agreement and the other transactions contemplated by this Agreement.
- (c) The Company and the Board of Directors of the Company have taken and will maintain in effect all necessary action to render the Company Rights Agreement inapplicable with respect to the Merger and the other transactions contemplated by this Agreement.
- 4.6 Compliance with Applicable Laws. (a) Each of the Company and its Subsidiaries has in effect all Federal, state, local and foreign governmental approvals, authorizations, certificates, filings, franchises, licenses, notices, permits and rights, including all authorizations under Environmental Laws and Gaming Laws ("Permits"), necessary for it to own, lease or operate its properties and assets and to carry on its business as now conducted other than such Permits the absence of which would not, individually or in the aggregate, have a Material Adverse Effect with respect to the Company, and there has occurred no default under any such Permit other than such defaults which, individually or in the aggregate, would not have a Material Adverse Effect with

respect to the Company. Except as disclosed in the Company SEC Documents filed and publicly available prior to the date of this Agreement, the Company and the Company Significant Subsidiaries are in compliance with all Applicable Laws, except for such noncompliance which individually or in the aggregate would not have a Material Adverse Effect with respect to the Company. The preceding sentence of this Section 4.6 does not apply to matters specifically covered by Section 4.10, 4.11 or 4.6(b) through 4.6(d).

- (b) Each of the Company and its Subsidiaries is, and has been, and each of the Company's former Subsidiaries, while a Subsidiary of the Company, was in compliance with all applicable Environmental Laws, except for such noncompliance which, individually or in the aggregate, would not have a Material Adverse Effect with respect to the Company. The term "ENVIRONMENTAL LAWS" means any applicable Federal, state, local or foreign statute, ordinance, rule, regulation, Permit, judgment, order, decree, injunction or other legally binding authorization relating to: (A) Releases (as defined in 42 U.S.C. SECTION 9601(22)) or threatened Releases of Hazardous Material (as hereinafter defined) into the environment or (B) the generation, treatment, storage, disposal, use, handling, manufacturing, transportation or shipment of, or exposure to, a Hazardous Material.
- (c) During the period of ownership or operation by the Company and its Subsidiaries of any of their owned or leased properties, there have been no Releases of Hazardous Material in, on, under or affecting such properties and none of the Company or its Subsidiaries have disposed of any Hazardous Material or any other substance in a manner that has led to, or could reasonably be anticipated to lead to, a Release except in each case for those which, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect with respect to the Company. The term "HAZARDOUS MATERIAL" means (1) hazardous substances (as defined in 42 U.S.C. SECTION 9601(14)). (2) petroleum, including crude oil and any fractions thereof, (3) natural gas, synthetic gas and any mixtures thereof, (4) asbestos and/or asbestos-containing material, (5) PCBs, or materials containing PCBs in excess of 50 ppm, and any material regulated as a medical waste or infectious waste.
- (d) The transactions contemplated by this Agreement will not require compliance with the New Jersey Industrial Site Recovery Act or any similar state transfer law.
- 4.7 Company SEC Documents; Undisclosed Liabilities. (a) Each of the Company and its Subsidiaries has filed all required reports, registration statements, proxy statements, forms and other documents with the SEC since January 1, 1995 (as such documents have since the time of their filing been amended or supplemented and, together with all reports, registration statements, forms and other documents filed by GNOC Corp., Bally's Park Place, Inc., Bally's Grand, Inc. and Bally's Casino Holdings, Inc. since January 1, 1995, the "Company SEC Documents"). As of their respective dates, (i) the Company SEC Documents (including any financial statements filed as a part thereof or incorporated by reference therein) complied in all material respects with the requirements of the Securities Act or the Exchange Act, as applicable, and the rules and regulations of the SEC thereunder applicable to such Company SEC Documents, and (ii) none of the Company SEC Documents contained at the time they were filed any untrue statement of a material fact or omitted at the time they were filed to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. At their respective dates, the financial statements of the Company included in the Company SEC Documents complied as to form in all material respects with applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto, were prepared in accordance with GAAP (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC) applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly presented (subject, in the case of the unaudited financial statements, to normal, recurring audit adjustments) the consolidated financial position of the Company and its consolidated Subsidiaries as at the dates thereof and the consolidated results of their operations and cash flows for the periods then ended.
- (b) Except as disclosed in the Company SEC Documents filed and publicly available prior to the date of this Agreement or in Section 4.8 or 6.1 of this Agreement or the related Sections of the Company Disclosure Schedule and except for liabilities and obligations incurred in the ordinary course of business consistent with past practice since December 31, 1995, the Company and its Subsidiaries do not have any material indebtedness,

obligations or liabilities of any kind (whether accrued, absolute, contingent or otherwise) (i) required by GAAP to be reflected on a consolidated balance sheet of the Company and its consolidated Subsidiaries or in the notes, exhibits or schedules thereto or (ii) which reasonably could be expected to have a Material Adverse Effect with respect to the Company.

4.8 Absence of Changes or Events. Except as disclosed in the Company SEC Documents filed and publicly available prior to the date of this Agreement or as set forth in Section 4.8 or 6.1 of the Company Disclosure Schedule or permitted by Section 6.1 of this Agreement, since December 31, 1995, the Company and its Subsidiaries have conducted their respective businesses only in the ordinary course, and there has not been (i) any change or occurrence which resulted in or is reasonably likely to have a Material Adverse Effect with respect to the Company, (ii) any declaration, setting aside or payment of any dividend or other distribution with respect to the capital stock of the Company other than on the Series D Preferred Stock or on the PRIDES, as specifically provided by the terms thereof, (iii) any issuance of any shares of Company Common Stock or other capital stock of the Company or any securities convertible into or exchangeable or exercisable for capital stock of the Company that is not reflected in Section 4.2 (other than issuances of Company Common Stock described in Section 6.1(ii)), (iv) any split, combination or reclassification of any of the capital stock of the Company or any issuance or the authorization of any issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock of the Company, (v) (x) any granting by the Company or any of its Subsidiaries to any director or officer of the Company or any of its Subsidiaries of any increase in compensation, except in the ordinary course of business consistent with prior practice, or as was required under employment agreements in effect as of December 31, 1995 that were included as an exhibit to a Company SEC Document filed and publicly available prior to the date of this Agreement, (y) any granting by the Company or any of its Subsidiaries to any such person of any increase in severance or termination pay, except as part of a standard employment package to any person promoted or hired (but not including the five most highly compensated senior officers of the Company and its Subsidiaries), or as was required under employment, severance or termination agreements in effect as of December 31, 1995 that were included as an exhibit to a Company SEC Document filed and publicly available prior to the date of this Agreement or as disclosed in Section 4.8 or 6.1 of the Company Disclosure Schedule or, (z) except for employment, severance or termination arrangements in the ordinary course of business consistent with past practice with employees other than any executive officer of the Company, any entry by the Company or any of its Subsidiaries into any employment, severance or termination agreement with any such person, (vi) other than those listed on Section 4.8 or 6.1 of the Company Disclosure Schedule, any acquisition of or commitment to purchase or build any property or project involving an expenditure in excess of \$10 million in the aggregate, (vii) any damage, destruction or loss not covered by insurance, that has or reasonably could be expected to have a Material Adverse Effect with respect to the Company or (viii) any change in accounting methods, principles or practices by the Company materially affecting its assets, liabilities or business, except insofar as may have been required by a change in GAAP.

4.9 Litigation. Except as disclosed in Section 4.9 of the Company Disclosure Schedule or in the Company SEC Documents filed and publicly available prior to the date of this Agreement, there are no Legal Proceedings pending against the Company or any of its Subsidiaries or, to the knowledge of the Company, threatened that, individually or in the aggregate, could reasonably be expected to (i) have a Material Adverse Effect with respect to the Company or (ii) prevent, or significantly delay the consummation of the transactions contemplated by this Agreement. Except as disclosed in Section 4.9 of the Company Disclosure Schedule or as set forth in the Company SEC Documents filed and publicly available prior to the date of this Agreement, there is no judgment, order, injunction or decree of any Governmental Authority outstanding against the Company or any of its Subsidiaries that, individually or in the aggregate, could reasonably be expected to have any effect referred to in the foregoing clauses (i) and (ii).

4.10 Taxes. (a) The Company and each of its Subsidiaries, and each affiliated group (within the meaning of Section 1504 of the Code) of which the Company or any of its Subsidiaries is or has ever been a member, has timely filed all Federal income Tax Returns and all other material Tax Returns and reports required to be filed by it. All such Tax Returns are complete and correct in all material respects. The Company and each of its

Subsidiaries has paid (or the Company has paid on its Subsidiaries' behalf) all taxes shown due on such Tax Returns. The most recent consolidated financial statements contained in the Company SEC Documents reflect an adequate reserve for all Taxes payable by the Company and its Subsidiaries for all taxable periods and portions thereof through the date of such financial statements.

- (b) Except as disclosed on Section 4.10 of the Company Disclosure Schedule, no material deficiencies for any Taxes have been proposed, asserted or assessed against the Company or any of its Subsidiaries that have not been fully paid or adequately provided for in the appropriate financial statements of the Company and its Subsidiaries, no requests for waivers of the time to assess any Taxes are pending, and no power of attorney with respect to any Taxes has been executed or filed with any taxing authority. No material issues relating to Taxes have been raised in writing by the relevant taxing authority during any presently pending audit or examination. The Federal income Tax Returns of the Company and each of its Subsidiaries consolidated in such Tax Returns have been reviewed with the Internal Revenue Service for all years through 1993.
- (c) No material liens for Taxes exist with respect to any assets or properties of the Company or any of its Subsidiaries, except for statutory liens for Taxes not yet due.
- (d) Except as disclosed on Section 4.10 of the Company Disclosure Schedule and other than with respect to contractual tax indemnity obligations of the Company and its Subsidiaries involving claims for state and local Taxes which are not material in amount, none of the Company or any of its Subsidiaries is a party to or is bound by any tax sharing agreement tax indemnity obligation or similar agreement, arrangement or practice with respect to Taxes (including any advance pricing agreement, closing agreement or other agreement relating to Taxes with any taxing authority).
- (e) None of the Company or any of its Subsidiaries has taken or agreed to take any action that would prevent the Merger from constituting a reorganization qualifying under the provisions of Section 368(a)(1) of the Code.
- (f) Except as disclosed in Section 4.10 of the Company Disclosure Schedule, there are no employment, severance or termination agreement, other compensation arrangement or Benefit Plan currently in effect which provide for the payment of any amount (whether in cash or property or the vesting of property) as a result of any of the transactions contemplated by this Agreement to any employee, officer or director of the Company or any of its affiliates who is a "disqualified individual" (as such term is defined in proposed Treasury Regulation Section 1.280G-1), that would be characterized as an "excess parachute payment" (as such term is defined in Section 280G(b)(1) of the Code).
- (g) The Company and it Subsidiaries have complied in all material respects with all applicable laws, rules and regulations relating to the payment and withholding of Taxes.
- (h) Except as disclosed in Section 4.10 of the Company Disclosure Schedule, no Federal, state, local or foreign audits or other administrative proceedings or court proceedings are presently pending with regard to any Federal income or material state, local or foreign Taxes or Tax Returns of the Company or its Subsidiaries and neither the Company nor ary of its Subsidiaries has received a written notice of any pending audit or proceeding.
- (i) Neither the Company nor any of its Subsidiaries has agreed to or is required to make any adjustment under Section 481(a) of the Code.
- (j) Neither the Company nor any of its Subsidiaries has, with regard to any assets or property held or acquired by any of them, filed a consent to the application of Section 341(f) of the Code or agreed to have Section 341(f)(2) of the Code apply to any disposition of a subsection (f) asset (as such term is defined in Section 341(f)(4) of the Code) owned by the Company or any of its Subsidiaries.
- (k) No property owned by the Company or any of its Subsidiaries (i) is property required to be treated as being owned by another Person pursuant to the provisions of Section 168(f)(8) of the Internal Revenue Code of

- 1954, as amended and in effect immediately prior to the enactment of the Tax Reform Act of 1986; (ii) constitutes "tax exempt use property" within the meaning of Section 168(h)(1) of the Code; or (iii) is tax exempt bond financed property within the meaning of Section 168(g) of the Code.
- 4.11 Employee Benefits. (a) Section 4.11(a) of the Company Disclosure Schedule lists all "employee benefit plans," as defined in Section 3(3) of ERISA and all other employee benefit plans or other benefit arrangements, including executive compensation and directors' benefit plans, and payroll practices which the Company or any of its Subsidiaries maintains, contributes to or has any obligation to or liability for (each an "Employee Benefit Plan" and collectively, the "Employee Benefit Plans").
- (b) Copies or descriptions of each Employee Benefit Plan (and, where applicable, the most recent summary plan description, actuarial report, determination letter, most recent Form 5500 and trust agreement) have been made available to Acquiror for review prior to the date hereof.
- (c) As of the date hereof, except as disclosed on Section 4.11(c) of the Company Disclosure Schedule, (i) all material payments required to be made by or under any Employee Benefit Plan, any related trusts, or any collective bargaining agreement have been made or are being processed in accordance with normal operating procedures, and except as set forth in the Company's financial statements, all material amounts required to be reflected thereon have been properly accrued to date as liabilities under or with respect to each Employee Benefit Plan for the current year; (ii) the Company and its Subsidiaries have performed all material obligations required to be performed by them under any Employee Benefit Plan; (iii) the Employee Benefit Plans, have been administered in material compliance with their terms and the requirements of ERISA, the Code and other Applicable Laws; (iv) there are no material actions, suits, arbitrations or claims (other than routine claims for benefit) pending or threatened with respect to any Employee Benefit Plan; and (v) the Company and its Subsidiaries have no liability as a result of any "prohibited transaction" (as defined in Section 406 of ERISA and Section 4975 of the Code) for any material excise tax or civil penalty.
- (d) Except as disclosed on Section 4.11(d) of the Company Disclosure Schedule, none of the Employee Benefit Plans which are "employee pension benefit plans" within the meaning of Section 3(2) of ERISA ("Pension Plans"), other than "multiemployer plans", as defined in Section 3(37) of ERISA ("Multiemployer Plans"), is subject to Title IV of ERISA.
- (e) Except as set forth on Section 4.11(e) of the Company Disclosure Schedule, the Company and its Subsidiaries have not, since April. 29, 1980, with respect to any Multiemployer Plan, suffered or otherwise caused a "complete withdrawal" or "partial withdrawal", as such terms are respectively defined in Sections 4023 and 4025 of ERISA which has resulted in any material liability to the Company or any of its Subsidiaries which has not been fully satisfied or which is not set forth in the Company's financial statements.
- (f) Except as set forth on Section 4.11(f) of the Company Disclosure Schedule, each of the Pension Plans which is intended to be "qualified" within the meaning of Section 401(a) of the Code has been determined by the Internal Revenue Service to be so "qualified" and the Company knows of no fact which would adversely affect the qualified status of any such Fension Plan.
- (g) Except as set forth on Section 4.11(g) of the Company Disclosure Schedule, neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (i) result in any material payment becoming due, or materially increase the amount of compensation due, to any current or former employee of the Company or any of its Subsidiaries; (ii) materially increase any benefits otherwise payable under any Employee Benefit Plan: or (ii) result in the acceleration of the time of payment or vesting of any such material benefits.
- 4.12 Title to Properties. (a) Section 4.12 of the Company Disclosure Schedule sets forth a complete list of all material real property owned in fee by the Company or one of its Subsidiaries and sets forth all material real property leased by the Company or one of its Subsidiaries as lessee as of the date hereof (such owned and leased

material real property, including all Improvements, referred to collectively as the "Real Property"). Except as set forth in Section 4.12 of the Company Disclosure Schedule, each of the Company and its Subsidiaries has good and valid title to, or a valid leasehold interest in, the Real Property held by it. Except as set forth in Section 4.12 of the Company Disclosure Schedule, the Real Property is free of Encumbrances, except for Permitted Encumbrances, and the consummation of the transactions contemplated by this Agreement will not create any Encumbrance on any of the Real Property which, individually or in the aggregate, would have a Material Adverse Effect with respect to the Company. Each of the Company and each of its Subsidiaries enjoys peaceful and undisturbed possession under all leases of Real Property, expect for such breaches of the right to peaceful and undisturbed possession that do not materially interfere with the ability of the Company and its Subsidiaries to conduct their business.

- (b) Except as set forth in Schedule 4.12 or as disclosed in the Company SEC Documents, no toxic or hazardous wastes, substances or materials are stored or otherwise held in violation of Environmental Laws that would have or could reasonably be expected to have a Material Adverse Effect with respect to the Company, and to the knowledge of the Company, no Contamination exists, on or under the Real Property at levels that contravene those allowed by Environmental Laws.
- 4.13 Insurance. The Company and its Subsidiaries have insurance coverage with insurance companies or associations in such amounts, on such terms and covering such risks, including fire and other risks insured against by extended coverage, as is reasonably prudent, and each has public liability insurance, insurance against claims for personal injury or death or property damage occurring in connection with any of activities of the Company or any of its Subsidiaries or of any properties owned, occupied or controlled by the Company or any of its Subsidiaries, in such amount as is deemed reasonably necessary by the Company or any of its Subsidiaries.
- 4.14 Brokers and Intermediaries. No broker, investment banker, financial advisor or other person, other than Goldman, Sachs & Co. and Merrill Lynch & Co. (whose fee arrangements have been disclosed to Acquiror in writing and will not be modified subsequent to the date of this Agreement), the fees and expenses of which will be paid by the Company, is entitled to any broker's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Company.
- 4.15 Opinion of Financial Advisor. The Company has received (i) the opinion of Goldman, Sachs & Co. to the effect that, as of the date of this Agreement, the consideration to be received in the Merger by the Company's stockholders is fair to the Company's stockholders, and (ii) the opinion of Merrill Lynch & Co. to the effect that, as of the date of this Agreement, the consideration to be received in the Merger by the Company's stockholders is fair to the Company's stockholders from a financial point of view.
- 4.16 Company Fights Agreement. The entering into of this Agreement and the consummation of the transactions contemplated hereby do not and will not result in the grant of any rights to any person under the Company Rights Agreement or enable or require the Company Rights to be exercised, distributed or triggered.
- 4.17 Transactions With Affiliates. Other than the transactions contemplated by this Agreement and except to the extent disclosed in the Company SEC Documents or as set forth in Schedule 4.17 of the Company Disclosure Schedule, from January 1, 1994 through the date of this Agreement, there have been no transactions, agreements, arrangements or understandings between the Company or its Subsidiaries, on the one hand, and the Company's affiliates (other than wholly-owned Subsidiaries of the Company and Bally's Grand, Inc.) or other Persons, on the other hand, that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act.

#### ARTICLE V

#### REPRESENTATIONS AND WARRANTIES OF ACQUIROR

Acquiror hereby represents and warrants to the Company as follows:

- 5.1 Organization, Standing and Corporate Power of Acquiror. Each of Acquiror and each Acquiror Significant Subsidiary is a corporation or partnership duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is organized and has the requisite corporate or partnership power and authority to carry on its business as now being conducted. Each of Acquiror and its Subsidiaries is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the nature of its business or the ownership or leasing of its properties makes such qualifications or licensing necessary, other than in such jurisdictions where the failure to be so qualified or licensed (individually or in the aggregate) would not have a Material Adverse Effect with respect to Acquiror. Acquiror has delivered to the Company complete and correct copies of its Restated Certificate of Incorporation and By-laws in each case as amended to the date of this Agreement. The respective certificates of incorporation and by-laws or other organizational documents of Acquiror Significant Subsidiaries do not contain any provision limiting or otherwise restricting the ability of Acquiror to control such Subsidiaries.
- 5.2 Capitalization. As of the date of this Agreement, the authorized capital stock of Acquiror consists of 90,000,000 shares of Acquiror Common Stock and 10,000,000 shares of preferred stock, par value \$1.00 per share (the "Acquiror Preferred Stock"). As of May 31, 1996, (i) 48,838,694 shares of Acquiror Common Stock and no shares of Acquiror Preferred Stock were issued and outstanding, (ii) 2,186,334 shares of Acquiror Common Stock were held by Acquiror in its treasury, (iii) 1,500,000 shares of Acquiror Common Stock were reserved for issuance upon the exercise of outstanding options issued under the Acquiror's 1996 Stock Incentive Plan, (iv) 1,500,000 shares of Acquiror Common Stock were reserved for issuance upon the exercise of outstanding options issued under Acquiror's 1996 Chief Executive Stock Incentive Plan, (v) 1,870,953 shares of Acquiror Common Stock were reserved for issuance upon the exercise of outstanding options issued under Acquiror's 1990 Stock Option and Stock Appreciation Rights Plan, (vi) 201,592 shares of Acquiror Common Stock were reserved for issuance upon the exercise of outstanding options issued under Acquiror's 1984 Stock Option and Stock Appreciation Fights Plan (the options issued pursuant to the stock option plans referred to in clauses (iii) through (vi) being collectively referred to as the "Acquiror Stock Options"), and (vii) 3,872,216.84 shares were reserved for issuance upon conversion of Acquiror's outstanding Convertible Subordinated Notes due 2006, which were presently convertible into 3,872,216.84 shares of Acquiror Common Stock. Except as set forth above, as of May 31, 1996, no shares of capital stock or other voting securities of Acquiror were issued, reserved for issuance or outstanding. As of the date of this Agreement, there are no outstanding stock appreciation rights which were rot granted in tandem with (and terminate upon exercise of) a related Acquiror Stock Option and there are no other outstanding contractual rights the value of which is derived from the financial performance of Acquiror or the value of shares of Acquiror Common Stock. All outstanding shares of capital stock of Acquiror are, and all shares which may be issued as contemplated by this Agreement will be, when issued, duly authorized, validly issued, fully paid and nonassessable and not subject to preemptive rights. As of the date of this Agreement, other than as set forth above, there are no bonds, debentures, notes or other indebtedness of Acquiror having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which stockholders of Acquiror may vote. Except as set forth above, or as disclosed in the Acquiror SEC Documents, as of the date of this Agreement, there are no outstanding securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind to which Acquiror or any of its Subsidiaries is a party or by which any of them is bound, obligating Acquiror or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other voting securities of Acquiror or of any of its Subsidiaries or obligating Acquiror or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking. As of the date of this Agreement, except as disclosed in the Acquiror SEC Documents or Section 5.2 of the Acquiror Disclosure Schedule, there are not any outstanding contractual obligations of Acquiror or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of Acquiror or any of its Subsidiaries.

A-19

- 5.3 Authority; Enforceability; No Conflicts. (a) Acquiror has the requisite corporate power and authority to enter into this Agreement and, subject to obtaining Acquiror Stockholder Approval, to consummate the transactions contemplated by this Agreement. The execution and delivery of this Agreement by Acquiror and the consummation by Acquiror of the transactions contemplated by this Agreement have been duly authorized by all necessary corporate action on the part of Acquiror, subject, in the case of the issuance of Acquiror Common Stock and New PRIDES contemplated hereby, to approval of such issuance by the holders of a majority of the votes cast at a special meeting of the holders of Acquiror Common Stock, provided that the total number of votes cast represents over 50% of the outstanding Acquiror Common Stock (the "Acquiror Stockholder Approval"). This Agreement has been duly executed and delivered by Acquiror and, assuming this Agreement constitutes a valid and binding obligation of the Company, constitutes valid and binding obligations of each of Acquiror, enforceable against Acquiror in accordance with its terms.
- (b) The execution and delivery of this Agreement do not, and the consummation of the transactions contemplated by this Agreement and compliance with the provisions of this Agreement will not, conflict with, or result in any violation of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation, or acceleration of any obligation or cause loss of a material benefit under, or result in the creation or maturation of any lien or purchase right upon any of the properties or assets of Acquiror and Acquiror Significant Subsidiaries under, (i) the certificate of incorporation or by-laws of Acquiror or the comparable charter or organizational documents of any of its Subsidiaries, (ii) except as set forth in Section 5.3 of the Acquiror Disclosure Schedule and subject to the governmental filings and other matters referred to in Section 5.3(c) any loan or credit agreement, note, bond, mortgage, indenture, lease or other agreement, instrument, permit, concession, franchise or license applicable to Acquiror or (iii) subject to the governmental filings and other matters referred to in Section 5.3(c), any judgment, order, decree, statute, law, ordinance, rule or regulation applicable to Acquiror or any of Acquiror Significant Subsidiaries or their respective properties or assets, other than, in the case of clauses (ii) or (iii), any such conflicts, violations, defaults, rights or liens that individually or in the aggregate would not (x) have a Material Adverse Effect with respect to Acquiror, (y) impair, in any material respect, the ability of Acquiror to perform its obligations under this Agreement or (z) prevent or significantly delay the consummation of any of the transactions contemplated by this Agreement.
- (c) No consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Authority is required by Acquiror or any of its Subsidiaries in connection with the execution and delivery of this Agreement or the consummation by Acquiror of any of the transactions contemplated by this Agreement, except for (i) the filing of a premerger notification and report form by Acquiror under the HSR Act, (ii) the filing with the SEC of (x) the Proxy Statement/Prospectus, (y) the Registration Statement on Form S-4 in connection with the issuance of Acquiror Common Stock and New PRIDES in the Merger (the "Form S-4") and the obtaining of any related orders as may be so required and (z) such reports and filings under Section 13 and Section 16 of the Exchange Act as may be required in connection with this Agreement and the transactions contemplated by this Agreement, (iii) the filing of the Certificate of Merger with the Delaware Secretary of State and appropriate documents with the relevant authorities of other states in which the Company is qualified to do business, (iv) the approval by (A) the New Jersey Casino Control Commission under the New Jersey Gaming Laws, (B) the Nevada State Gaming Control Board and the Nevada Gaming Commission under the Nevada Gaming Laws, (C) the Clark County Liquor and Gaming Licensing Board pursuant to the Clark County, Nevada Code and the rules and regulations promulgated thereunder, (D) the Mississippi Gaming Commission under the Mississippi Gaming Laws and (E) the Louisiana Riverboat Gaming Commission under the Louisiana Gaming Laws, (v) the filing of notices and the approval of Governmental Authorities as may be required under the applicable Gaming Laws of the States of Maryland and Washington and under the Indian Gaming Regulatory Act of 1988, (vi) as may be required by any applicable state securities or "blue sky" laws, and (vii) such other consents, approvals, orders, authorizations, registrations, declarations and filings the failure of which to be obtained or made would not, individually or in the aggregate, (x) have a Material Adverse Effect with respect to Acquiror, (y) impair, in any material respect, the ability of Acquiror to perform its obligations under this Agreement or (z) prevent or significantly delay the consummation of the transactions contemplated by this Agreement.

- 5.4 The Vote Required; Ownership of Company Capital Stock. (a) Acquiror Stockholder Approval is the only vote of the holders of any class or series of Acquiror's capital stock necessary to approve this Agreement and the transactions contemplated hereby.
- (b) Except for two shares of Company Common Stock, neither Acquiror nor any of its Subsidiaries owns, directly or indirectly, any shares of Company Capital Stock.
- 5.5 Acquiror SEC Documents; Undisclosed Liabilities. (a) Acquiror has filed all required reports, registration statements, proxy statements, forms and other documents with the SEC since January 31, 1995 (as such documents have since the time of their filing been amended or supplemented, the "Acquiror SEC Documents"). As of their respective dates, (i) Acquiror SEC Documents (including any financial statements filed as a part thereof or incorporated by reference therein) complied in all material respects with the requirements of the Securities Act or the Exchange Act, as applicable, and the rules and regulations of the SEC promulgated thereunder applicable to such Acquiror SEC Documents, and (ii) none of Acquiror SEC Documents contained at the time they were filed any untrue statement of a material fact or omitted at the time they were filed to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. At their respective dates, the financial statements of Acquiror included in Acquiror SEC Documents complied as to form in all material respects with applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto, were prepared in accordance with GAAP (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC) applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly presented (subject, in the case of unaudited statements, to normal year-end audit adjustments) the consolidated financial position of Acquiror and its consolidated subsidiaries as at the dates thereof and the consolidated results of their operations and cash flows for the periods then ended.
- (b) Except as disclosed on Section 5.5 of Acquiror Disclosure Schedule or in the Acquiror SEC Documents filed and publicly available after December 31, 1995 and prior to the date of this Agreement and except for liabilities and obligations incurred in the ordinary course of business consistent with past practice since December 31, 1995, as of the date of this Agreement Acquiror and its Subsidiaries do not have any material indebtedness, obligations or liabilities of any kind (whether accrued, absolute, contingent or otherwise) required by GAAP to be reflected on a consolidated balance sheet of Acquiror and its consolidated Subsidiaries or in the notes, exhibits or schedules thereto.
- 5.6 Absence of Changes or Events. Except as disclosed in the Acquiror SEC Documents filed and publicly available prior to the date of this Agreement or in the 1995 Acquiror Financial Statements, (i) since December 31, 1995, there has not been any change or occurrence which resulted in or is reasonably likely to have a Material Adverse Effect with respect to Acquiror, and (ii) from December 31, 1995 to the date of this Agreement, Acquiror and its Subsidiaries have conducted their businesses only in the ordinary course and there have not been (A) any declaration, setting aside or payment of any dividend or other distribution with respect to the capital stock of Acquiror, other than regular quarterly dividends on Acquiror Common Stock, (B) any split, combination or reclassification of any of the capital stock of Acquiror or any issuance or the authorization of any issuance of any other securities in lieu of or in substitution for shares of capital stock of Acquiror, (C) any damage, destruction or loss not covered by insurance, that has or reasonably could be expected to have a Material Adverse Effect with respect to Acquiror or (D) any change in accounting methods, principles or practices by Acquiror materially affecting its assets, liabilities or business, except insofar as may have been disclosed by a change in generally accepted accounting principles.
- 5.7 Litigation. Except as disclosed in the Acquiror SEC Documents filed and publicly available prior to the date of this Agreement, there are no Legal Proceedings pending against Acquiror or any of its Subsidiaries or, to the knowledge of Acquiror, threatened that, individually or in the aggregate, could reasonably be expected to (i) have a Material Adverse Effect with respect to Acquiror or (ii) prevent or significantly delay the consummation of any of the transactions contemplated by this Agreement. Except as disclosed in the Acquiror SEC Documents filed and publicly available prior to the date of this Agreement, there is no judgment, order,

injunction or decree of any Governmental Authority outstanding against Acquiror or any of its Subsidiaries that, individually or in the aggregate, could reasonably be expected to have any effect referred to in the foregoing clauses (i) and (ii).

- 5.8 Taxes. (a) Acquiror and each of its Subsidiaries, and each affiliated group (within the meaning of Section 1504 of the Code) of which the Company or any of its Subsidiaries is or has ever been a member, has timely filed all Federal income tax returns and all other material tax returns and reports required to be filed by it. All such returns are complete and correct in all material respects. Acquiror and each of its Subsidiaries has paid (or Acquiror has paid or its Subsidiaries' behalf) all taxes shown due on such returns, the most recent financial statements contained in the Acquiror SEC Documents reflect an adequate reserve for all Taxes payable by Acquiror and its Subsidiaries for all taxable periods and portions thereof through the date of such financial statements, and the 1995 Financial Statements reflect an adequate reserve for all Taxes payable by Acquiror and its Subsidiaries for all taxable periods and portions thereof through December 31, 1995.
- (b) Neither Acquirer nor any of its Subsidiaries has taken or agreed to take any action that would prevent the Merger from constituting a reorganization qualifying under the provisions of Section 368(a)(1) of the Code.
- 5.9 Compliance with Applicable Laws. (a) Each of Acquiror and its Subsidiaries has in effect all Permits necessary for it to own, lease or operate its properties and assets and to carry on its business as now conducted, and there has occurred no default under any such Permit, except for the lack of Permits and for defaults under Permits which lack or default, individually or in the aggregate, would not have a Material Adverse Effect with respect to Acquiror. Except as disclosed in the Acquiror SEC Documents filed and publicly available prior to the date of this Agreement, Acquiror and its Subsidiaries are in compliance with all Applicable Laws, except for possible noncompliance which, individually or in the aggregate, would not have a Material Adverse Effect with respect to Acquiror.
- (b) Except as disclosed in the Acquiror SEC Documents, each of Acquiror and its Subsidiaries is, and has been, and each of Acquiror's former Subsidiaries, while a Subsidiary of Acquiror was in compliance with all applicable Environmental Laws, except for possible noncompliance which, individually or in the aggregate, would not have a Material Adverse Effect with respect to the Company.
- (c) Except as disclosed in the Acquiror SEC Documents, during the period of ownership or operation by Acquiror and its Subsidiaries of any of their owned or leased properties, there have been no Releases of Hazardous Material in, on under or affecting such properties and none of Acquiror or its Subsidiaries have disposed of any Hazardous Material or any other substance in a manner that has led to, or could reasonably be anticipated to lead to, a Release except in each case for those which, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect with respect to Acquiror.
- 5.10 ERISA Compliance. Except as described in Acquiror SEC Documents filed and publicly available prior to the date of this Agreement or as would not have a Material Adverse Effect with respect to Acquiror, (i) all employee benefit plans or programs maintained for the benefit of the current or former employees or directors of Acquiror or any Subsidiary of Acquiror that are sponsored, maintained or contributed to by Acquiror or any Subsidiary of Acquiror, or with respect to which Acquiror or any Subsidiary of Acquiror has any liability, including any such plan that is an "employee benefit plan" as defined in Section 3(3) of ERISA, are in compliance with all applicable requirements of law, including ERISA and the Code, and (ii) neither Acquiror nor any Subsidiary of Acquiror has any liabilities or obligations with respect to any such employee benefit plans or programs, whether accrued, contingent or otherwise, except liabilities or obligations incurred in the ordinary course.
- 5.11 Brokers. No broker, investment banker, financial advisor or other person, other than Donaldson, Lufkin & Jenrette Securities Corporation, the fees and expenses of which will be paid by Acquiror, is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Acquiror.

#### ARTICLE VI

## COVENANTS RELATING TO CONDUCT OF BUSINESS

- 6.1 Conduct of Business of the Company. Except as otherwise provided by the terms of this Agreement or as set forth in Section 4.8 of this Agreement or in Section 4.8 or 6.1 of the Company Disclosure Schedule, from and after the date hereof to the Effective Time, the Company shall, and shall cause its Subsidiaries to, carry on their respective businesses in the ordinary course and use their reasonable efforts to preserve intact their current business organizations, keep available the services of their current officers and key employees and preserve their relationships consistent with past practice with desirable customers, suppliers and others having business dealings with them to the end that their goodwill and ongoing businesses shall be unimpaired in all material respects at the Effective Time. Without limiting the generality of the foregoing, prior to the Effective Time, except as otherwise provided by the terms of this Agreement or as set forth in Section 4.8 of this Agreement or in Section 4.8 or 6.1 of the Company Disclosure Schedule, the Company shall not (and shall cause its Subsidiaries not to), without the written consent of Acquiror, which consent may not be unreasonably withheld:
  - (i) (A) declare, set aside or pay any dividends on, or make any other distributions in respect of, any of its capital stock, other than dividends and distributions by any direct or indirect wholly-owned subsidiary of the Company to its parent and other than as specifically provided in the terms of the PRIDES, the Series D Preferred and the Subsidiary Preferred Stock, (B) split, combine or reclassify any of its capital stock or, except pursuant to the exercise of options, warrants, conversion rights, exchange rights and other contractual rights existing on the date hereof, issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock or other equity interests or (C) purchase, redeem or otherwise acquire or amend any shares of capital stock or other equity interests of the Company or any of its Subsidiaries or any other securities thereof or any rights, warrants or options to acquire any such shares, interests or other securities (other than (x) redemptions, purchases or other acquisitions required by applicable provisions under Gaming Laws or pursuant to the terms of such capital stock or equity interest or other contractual rights existing on the date hereof, (y) issuances or redemptions of capital stock of wholly-owned Subsidiaries occurring between the Company and any of its wholly-owned Subsidiaries or occurring between wholly-owned Subsidiaries of the Company and (z) issuances to or redemptions from unaffiliated third parties of capital stock of or ownership interests in Subsidiaries not currently conducting significant operations and organized for purposes of pursuing new business opportunities or developing new properties;
  - (ii) issue, deliver, sell, pledge or otherwise encumber or amend any shares of its capital stock, any other voting securities or any securities convertible into, or any rights, warrants or options to acquire, any such shares, interests, voting securities or convertible securities, including pursuant to the Company's Employee Stock Purchase Flan (other than (A) the issuance of Company Common Stock upon the conversion of the PRIDES or Series D Preferred outstanding on the date of this Agreement in accordance with their present terms, (B) the issuance of Company Common Stock upon the conversion of Convertible Debentures outstanding on the date of this Agreement in accordance with their present terms, (C) the issuance of Company Common Stock upon the exercise of Options outstanding on the date of this Agreement in accordance with their present terms, (D) the issuance of Company Common Stock in the Subsidiary Merger pursuant to Section 7.16; and (E) the issuances described in subclauses (y) and (z) of paragraph (i) above);
  - (iii) amend its Restated Certificate of Incorporation, by-laws or other comparable charter or organizational documents;
  - (iv) acquire or agree to acquire (A) by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, joint venture, association or other business organization or division thereof or (B) any assets that are material, individually or in the aggregate, to the Company and its Subsidiaries taken as a whole, except (w) mergers and consolidations between or among one or more wholly-owned Subsidiaries of the Company that will not

create adverse tax consequences to the Company or its Subsidiaries, (x) purchases of inventory, furnishings and equipment in the ordinary course of business consistent with past practice, (y) the merger of Bally's Casino, Inc. with and into the Company pursuant to Section 7.16 hereof or (z) expenditures listed on Section 6.1 of the Company Disclosure Schedule;

- (v) sell, lease, license, mortgage or otherwise encumber or subject to any lien or otherwise dispose of any of its properties or assets, except in the ordinary course of business consistent with past practice;
- (vi) (A) other than (1) ordinary course working capital borrowings, (2) borrowings required to finance specific projects listed on Section 6.1 of the Company Disclosure Schedule, (3) other incurrences of indebtedness which, in the aggregate, do not exceed \$25 million, incur any indebtedness for borrowed money or guarantee any such indebtedness of another person, issue or sell any debt securities or warrants or other rights to acquire any debt securities of the Company or any of its Subsidiaries, guarantee any debt securities of another person, enter into any "keep well" or other agreement to maintain any financial statement condition of another person or enter into any arrangement having the economic effect of any of the foregoing or (B) make any loans, advances or capital contributions to, or investments in, any other person other than (v) to the Company or any direct or indirect wholly-owned Subsidiary of the Company, (w) advances to employees, suppliers or customers in the ordinary course of business consistent with past practice and (x) in connection with specific projects listed on Section 6.1 of the Company Disclosure Schedule;
- (vii) pay, discharge, settle or satisfy any claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), other than the payment, discharge, settlement or satisfaction, (A) in the ordinary course of business consistent with past practice, (B) in accordance with their terms of liabilities reflected or reserved against in the most recent consolidated financial statements (or the notes thereto) of the Company included in the Company SEC Documents filed and publicly available prior to the date of this Agreement or incurred in the ordinary course of business consistent with past practice since the date of such financial statements or (C) involving an amount not to exceed \$25 million in the aggregate, or waive the benefits of, or agree to modify in any manner, any confidentiality, standstill or similar agreement to which the Company or any of its Subsidiaries is a party;
- (viii) except as required to comply with Applicable Law, (A) adopt, enter into, terminate or amend any Employee Benefit Plan or other arrangement for the benefit or welfare of any director, officer or current or former employee, (B) increase in any manner the compensation or fringe benefits of, or pay any bonus to, any director, officer or employee (except for normal increases or bonuses as contractually required pursuant to agreements disclosed in the Company SEC Documents filed and publicly available prior to the date of this Agreement or in Section 4.8 of the Company Disclosure Schedule or in the ordinary course of business consistent with past practice to employees other than directors and executive officers of the Company and that, in the aggregate, do not result in a significant increase in benefits or compensation expense to the Company and its Subsidiaries relative to the level in effect prior to such action and except as contractually required pursuant to agreements included as part of a Company SEC Document filed and publicly available prior to the date of this Agreement), (C) pay any benefit not provided for under any Employee Benefi: Plan, (D) except for payments or awards in cash permitted by clause (B), grant any awards under any bonus, incentive, performance or other compensation plan or arrangement or Employee Benefit Plan (including the grant of stock options, stock appreciation rights, stock based or stock related awards, performance units or restricted stock, or the removal of existing restrictions in any Employee Benefit Plans or agreements or awards made thereunder) or (E) take any action to fund or in any other way secure the payment of compensation or benefits under any employee plan, agreement, contract or arrar gement or Employee Benefit Plan other than in the ordinary course of business consistent with past practice; provided, however, that the Company may take any action described in clauses (A), (B), (C) and (D) above in connection with the retention of employees and the payment of bonuses to executive officers of the Company (other than the Chairman of the Board, Chief Executive Officer and President of the Company) in an amount not to exceed 200% of their respective bonus compensation for fiscal year 1995 and that, taken together, has an aggregate economic cost to the Company not to exceed \$10,000,000;

- (ix) except in the ordinary course of business, modify, amend or terminate any contract or agreement set forth in the Company SEC Documents to which the Company or any Subsidiary is a party or waive, release or assign any material rights or claims;
- (x) take or agree to take any action that would prevent the Merger from constituting a reorganization qualifying under the provisions of Section 368(a)(1) of the Code;
- (xi) conduct its business in a manner or take, or cause to be taken, any other action that would or might reasonably be expected to prevent or materially delay the Company or Acquiror from consummating the transactions contemplated hereby in accordance with the terms of this Agreement (regardless of whether such action would otherwise be permitted or not prohibited hereunder), including, without limitation, any action which may materially limit the ability of the Company or Acquiror to consummate the transactions contemplated hereby as a result of antitrust, gaming or other regulatory concerns; or
  - (xii) authorize any of, or commit or agree to take any of, the foregoing actions.
- 6.2 Conduct of Business of Acquiror. Except as set forth in Section 6.2 of the Acquiror Disclosure Schedule, from and after the date hereof to the Effective Time, Acquiror shall, and shall cause its Subsidiaries to, carry on their respective current operations in the ordinary course and use their reasonable efforts to preserve intact their current business organizations, keep available the services of their current officers and key employees and preserve their relationships with desirable customers, suppliers and others having business dealings with them to the end that their goodwill and ongoing businesses shall be unimpaired in all material respects at the Effective Time. Without limiting the generality of the foregoing, prior to the Effective Time, except as specifically permitted by the terms of this Agreement, Acquiror shall not (and shall cause its Subsidiaries not to), without the written consent of the Company, which consent may not be unreasonably withheld:
  - (i) (A) declare, set aside or pay any dividends on, or make any other distributions in respect of Acquiror Common Stock, other than regular quarterly dividends, or (B) other than the stock split described in Section 3.1(d), split, combine or reclassify any of its Acquiror Common Stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for Acquiror Common Stock;
  - (ii) issue or authorize the issuance of any shares of Acquiror Common Stock or any option, warrant or right relating thereto or any securities convertible into or exchangeable for any shares of Acquiror Common Stock at less than fair market value per share of Acquiror Common Stock as determined by the Board of Directors of Acquiror (other than pursuant to the terms of existing options or benefit plans or convertible securities);
  - (iii) amend its Restated Certificate of Incorporation, by-laws or other comparable charter or organizational documents in any manner adverse to the holders of Acquiror Common Stock (other than the filing of a Certificate of Designations for the issuance of any series of Preferred Stock of Acquiror);
  - (iv) take or agree to take any action that would prevent the Merger from constituting a reorganization qualifying under the provisions of Section 368(a)(1) of the Code;
  - (v) conduct its business in a manner or take, or cause to be taken, any other action (including, without limitation, effecting or agreeing to effect or announcing an intention or proposal to effect, any acquisition, business combination, merger, consolidation, restructuring or similar transaction) that would or might reasonably be expected to prevent or materially delay Acquiror or the Company from consummating the transactions contemplated hereby in accordance with the terms of this Agreement (regardless of whether such action would otherwise be permitted or not prohibited hereunder), including, without limitation, any action which may materially limit the ability of Acquiror or the Company to consummate the transactions contemplated hereby as a result of antitrust, gaming or other regulatory concerns;
  - (vi) sell all or substantially all of the properties and assets of Acquiror or merge, amalgamate or consolidate Acquiror with any other entity in any transaction in which Acquiror is not the surviving corporation; or
    - (vii) authorize any of, or commit or agree to take any of, the foregoing actions.

Nothing contained in this Section 6.2 shall prohibit Acquiror from acquiring or agreeing to acquire, whether by merger or consolidation or by purchase of assets, any business or any corporation, partnership, joint venture, association or other business organization or division thereof or interest therein.

6.3 Access to Information. Each of the Company and Acquiror shall, and shall cause each of its respective Subsidiaries to, afford to the other party and to the officers, employees, accountants, counsel, financial advisors and other representatives of such other party, reasonable access during normal business hours during the period prior to the Effective Time to all their respective properties, books, contracts, commitments, personnel and records and, during such period, each of the Company and the Acquiror shall, and shall cause each of its respective Subsidiaries to, furnish promptly to the other party (a) a copy of each report, schedule, registration statement and other document filed by it during such period pursuant to the requirements of Federal or state securities laws and (b) all other information concerning its business, properties and personnel as such other party may reasonably request. Except as required by Applicable Law, each of the Company and Acquiror will hold, and will cause its respective officers, employees, accountants, counsel, financial advisors and other representatives and affiliates to hold, any nonpublic information in confidence to the extent required by, and in accordance with, the provisions of the letter, dated March 11, 1996, between the Company and Acquiror (the "Confident ality Agreement").

## ARTICLE VII

#### ADDITIONAL AGREEMENTS

- 7.1 Preparation of Form S-4 and the Proxy Statement/Prospectus; Stockholders' Meeting. (a) Promptly following the date of this Agreement, the Company and Acquiror shall prepare and file with the SEC the Proxy Statement/Prospectus, and Acquiror shall prepare and file with the SEC the Form S-4, in which the Proxy Statement/Prospectus shall be included. Each of the Company and Acquiror shall use its reasonable best efforts to have the Form S-4 declared effective under the Securities Act as promptly as practicable after such filing. The Company and Acquiror shall each use reasonable best efforts to cause the Proxy Statement/Prospectus to be mailed to their respective stockholders, as promptly as practicable after the Form S-4 is declared effective under the Securities Act. Acquiror shall also take any action (other than qualifying to do business in any jurisdiction in which it is not now so qualified or consenting to service of process in any jurisdiction in any action other than one arising out of the offering of the Acquiror Common Stock in such jurisdiction) required to be taken under any applicable state securities or "blue sky" laws in connection with the issuance of shares of Acquiror Common Stock and New PRIDES in the Merger, and the Company shall furnish all information concerning the Company as may be reasonably requested in connection with any such action.
- (b) Each of the Company and the Acquiror covenants that none of the information supplied or to be supplied by it for inclusion or incorporation by reference in (i) the Form S-4 will, at the time the Form S-4 is filed with the SEC, at any time it is amended or supplemented or at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, (ii) the Proxy Statement/ Prospectus will, at the date it is first mailed to the stockholders of the Company or Acquiror, or at the time of the Company Stockholders' Meeting or the Acquiror Stockholders' Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. The Proxy Statement/ Prospectus and the Form S-4 will comply as to form in all material respects with the requirements of the Exchange Act or the Securities Act, as the case may be. Notwithstanding the foregoing, (i) no representation or covenant is made by the Company with respect to statements made or incorporated by reference therein based on information supplied in writing by Acquiror specifically for inclusion or incorporation by reference in the Proxy Statement/Prospectus and (ii) no representation or covenant is made by Acquiror with respect to statements made or incorporated by reference therein based on information supplied in writing by the Company for inclusion or incorporation by reference in the Proxy Statement/Prospectus. If at any time prior to the Effective Time there

A-26

- shall occur (i) any event with respect to the Company or any of its Subsidiaries, or with respect to other information supplied by the Company for inclusion in the Proxy Statement/Prospectus or (ii) any event with respect to Acquiror, or with respect to information supplied by Acquiror for inclusion in the Proxy Statement/Prospectus, in either case which event is required to be described in an amendment of, or a supplement to, the Proxy Statement/Prospectus or the Form S-4, such event shall be so described, and such amendment or supplement shall be promptly filed with the SEC and, as required by law, disseminated to the stockholders of the Company and to the stockholders of Acquiror.
- (c) Each of the Company and Acquiror shall promptly notify the other of the receipt of any comments from the SEC or its staff or any other appropriate government official and of any requests by the SEC or its staff or any other appropriate government official for amendments or supplements to any of the filings with the SEC in connection with the Merger and other transactions contemplated hereby or for additional information and shall supply the other with copies of all correspondence between the Company or any of its representatives, or Acquiror or any of its representatives, as the case may be, on the one hand, and the SEC or its staff or any other appropriate government official, on the other hand, with respect thereto. The Company and Acquiror shall use their respective reasonable efforts to respond to any comments of the SEC with respect to the Form S-4 as promptly as practicable. The Company and Acquiror shall cooperate with each other and provide to each other all information necessary in order to prepare the Form S-4 and the Proxy Statement/Prospectus, and shall provide promptly to the other parties any information such party may obtain that could necessitate amending any such document.
- (d) The Company shall take all action necessary in accordance with Applicable Law and its Restated Certificate of Incorporation and By-laws to convene and hold a meeting of its stockholders (the "Company Stockholders' Meeting") as promptly as practicable for the purpose of obtaining the Company Stockholder Approval. The Company shall, through its Board of Directors, recommend to its stockholders the adoption of this Agreement and the transactions contemplated hereby and shall use its reasonable best efforts to solicit from its stockholders proxies in favor of adoption of this Agreement and to take all other lawful action necessary to secure the Company Stockholder Approval. Notwithstanding the foregoing, the Company's obligation to convene and hold the Company Stockholders' Meeting and to recommend the adoption of this Agreement and to solicit proxies from its stockholders shall be subject to any action (including any withdrawal or change of its recommendation) taken by, or upon authority of, the Board of Directors of the Company which the Board of Directors determines, based on the advice of outside legal counsel to the Company, is required in the exercise of its fiduciary duties to the Company's stockholders under Applicable Law.
- (e) Acquiror shall take all action necessary in accordance with Applicable Law and its Certificate of Incorporation and By-laws to convene and hold a meeting of its stockholders (the "Acquiror Stockholders' Meeting") as promptly as practicable for the purpose of obtaining the Acquiror Stockholder Approval. Acquiror shall, through its Board of Directors, recommend to its stockholders that they approve such issuances and shall use its reasonable best efforts to solicit from its stockholders proxies in favor of approving such issuances and to take all other lawful action necessary to secure the Acquiror Stockholder Approval; provided, however, that such recommendation or solicitation is subject to any action (including any withdrawal or change of its recommendation) taken by, or upon authority of, the Board of Directors of Acquiror which the Board of Directors determines, based on the advice of outside legal counsel to Acquiror, is required in the exercise of its fiduciary duties to the Acquiror's stockholders under Applicable Law.
- (f) The Company and Acquiror shall coordinate and cooperate with each other with respect to the timing of the Company Stockholders' Meeting and the Acquiror Stockholders' Meeting and shall use their reasonable efforts to hold such meetings on the same day and as soon as practicable after the date hereof.
- 7.2 Letter of the Company's Accountants. The Company shall use its best efforts to cause to be delivered to Acquiror a letter of Ernst & Young LLP, the Company's independent public accountants, dated a date within two Business Days before the date on which the Form S-4 shall become effective and addressed to Acquiror, in form and substance reasonably satisfactory to Acquiror and customary in scope and substance for letters delivered by independent public accountants in connection with registration statements similar to the Form S-4.

- 7.3 Letter of Acquiror's Accountants. Acquiror shall use its best efforts to cause to be delivered to the Company a letter of Arthur Andersen LLP, Acquiror's independent public accountants, dated a date within two Business Days before the date on which the Form S-4 shall become effective and addressed to the Company, in form and substance reasonably satisfactory to the Company and customary in scope and substance for letters delivered by independent public accountants in connection with registration statements similar to the Form S-4.
- 7.4 Reasonable Best Efforts; Notification. (a) Upon the terms and subject to the conditions set forth in this Agreement, each of the parties agrees to use all reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Merger and the other transactions contemplated by this Agreement, including (i) the obtaining of all necessary actions or nonactions, waivers, consents and approvals from Governmental Authorities and the making of all necessary registrations and filings (ir cluding filings with Governmental Authorities, if any) and the taking of all reasonable steps as may be necessary to obtain an approval or waiver from, or to avoid an action or proceeding by, any Governmental Authority (including in respect of any Gaming Law), (ii) the obtaining of all necessary consents, approvals or waivers from third parties, (iii) the defending of any lawsuits or other legal proceedings, whether judicial or administrative, challenging this Agreement or the consummation of any of the transactions contemplated by this Agreement, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Authority vacated or reversed and (iv) the execution and delivery of any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement.
- (b) The Company shall give prompt notice to Acquiror, and Acquiror shall give prompt notice to the Company, of (i) any representation or warranty made by it contained in this Agreement becoming untrue or inaccurate in any material respect (including in the case of representations or warranties by the Company or Acquiror, as applicable, such party's receiving knowledge of any fact, event or circumstance which may cause any representation qualified as to the knowledge of such party to be or become untrue or inaccurate in any material respect) or (ii) the failure by it to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied by it under this Agreement; provided, however, that no such notification shall affect the representations, warranties, covenants or agreements of the parties or the conditions to the obligations of the parties under this Agreement.
- 7.5 Approval of Gaming Commissions; Regulatory Matters. Acquiror shall as promptly as practicable, but in no event later than fifteen Business Days following the execution and delivery of this Agreement, file or submit those filings and other submissions under applicable Gaming Laws in connection with the Merger, this Agreement and the transactions contemplated hereby, and to respond as promptly as practicable to inquiries received from state or local gaming authorities and to appear before such authorities as promptly as practicable in order to obtain as soon as practicable those approvals and consents required or necessary in connection with the Merger, this Agreement or the transactions contemplated hereby. In addition, Acquiror shall, and shall cause its Subsidiaries to (and shall use its reasonable efforts to cause its affiliates other than its Subsidiaries to), if it is necessary to obtain any regulatory approval for the Merger, disassociate themselves from any person or persons deemed, or reasonably likely to be deemed, unacceptable by any Gaming Commission and, in the case of any such person who is a nominee to serve as a director of Acquiror or any of its Subsidiaries, Acquiror shall, and shall cause its relevant Subsidiary or Subsidiaries to replace any such director nominee with a suitable nominee. Acquiror shall keep the Company apprised of the status of any communications with, and any inquiries or requests for additional information from, the Gaming Commissions and shall comply promptly with any such inquiry or request. Acquiror shall use its reasonable best efforts to obtain all required approvals of the Gaming Commissions for the consummation of the Merger no later than October 15, 1996.
- 7.6 Supplemental Disclosure. The Company shall confer on a regular and frequent basis with Acquiror, report on operational matters and promptly notify Acquiror of, and furnish Acquiror with, any information it may reasonably request with respect to, any event or condition or the existence of any fact that would cause any of the conditions to Acquiror's obligation to consummate the Merger not to be completed, and Acquiror shall

promptly notify the Company of, and furnish the Company any information it may reasonably request with respect to, any event or condition or the existence of any fact that would cause any of the conditions to the Company's obligation to consummate the Merger not to be completed.

- 7.7 Announcements. Prior to the Closing, neither the Company nor Acquiror will issue any press release or otherwise make any public statement with respect to this Agreement and the transactions contemplated hereby without the prior consent of the other (which consent shall not be unreasonably withheld), except as may be required by Applicable Law or applicable stock exchange regulations, in which event the party required to make the release or announcement shall, if possible, allow the other party reasonable time to comment on such release or announcement in advance of such issuance. The parties agree that the initial press release to be issued with respect to the transactions contemplated by this Agreement shall be in the form heretofore agreed to by the parties.
- 7.8 No Solicitation. (a) From and after the date hereof until the Effective Time, the Company shall not, nor shall it authorize any of its officers, directors, employees, agents, investment bankers, attorneys, financial advisors or other representatives (collectively, "Company Representatives") to (i) solicit, initiate or knowingly encourage the submission of, any Acquisition Proposal, (ii) enter into any agreement with respect to any Acquisition Proposal, or (iii) participate in any discussions or negotiations regarding, or furnish to any Person any non-public information with respect to, or take any other action to knowingly facilitate any inquiries or the making of any proposal that constitutes or would reasonably be expected to lead to, an Acquisition Proposal; provided, however, that, notwithstanding anything to the contrary in this Agreement, (i) the Company may participate in discussions or negotiations with, and may furnish information concerning the Company and its business, properties and assets to, a Third Party who, without any solicitation by the Company or any Company Representatives after the date of this Agreement, seeks to engage in such discussions or negotiations or requests such information, if (1) the Board of Directors of the Company determines, based on the advice of the Company's outside legal counsel, that failing to engage in such discussion or negotiations or provide such information would reasonably be expected to violate the fiduciary duties of the Board of Directors of the Company to its stockholders and (2) prior to engaging in discussions or negotiations with, or furnishing information to, such Third Party, the Company shall receive from such Third Party an executed confidentiality agreement in reasonably customary form on terms not more favorable to such Person or entity than the terms contained in the Confidentiality Agreement, and (ii) the Board of Directors of the Company may take and disclose to the Company's stockholders a position with regard to a tender offer or exchange offer contemplated by Rules 14d-9 and 14e-2(a) promulgated under the Exchange Act and may make such disclosure to the stockholders of the Company as may be required under Applicable Law; provided, that the Board of Directors of the Company shall not recommend that the stockholders of the Company tender their shares of Company Common Stock unless such recommendation is permitted by Section 7.8(d).
- (b) The Company shall notify Acquiror of any Acquisition Proposal, including the identity of the Third Party making any such Acquisition Proposal and the material terms and conditions of any Acquisition Proposal.
- (c) As used in this Agreement, "Acquisition Proposal" shall mean any proposal or offer from any person relating to (i) any direct or indirect acquisition or purchase of more than 20% of either the capital stock of the Company or the consolidated assets of the Company and its Subsidiaries taken as a whole, (ii) any tender offer or exchange offer that if consummated would result in any person beneficially owning 20% or more of the capital stock of the Company or (iii) any merger, consolidation or business combination, involving the Company other than the transactions contemplated by this Agreement; provided, however, that any acquisition of shares of capital stock of the Company by FMR Corp. (and its subsidiaries, Fidelity Management & Research Company and Fidelity Management Trust Company) described in clause (i) shall not constitute an Acquisition Proposal.
- (d) Notwithstanding anything to the contrary in this Agreement, the Board of Directors of the Company shall be permitted from time to time to take the following actions in the circumstances described below: (i) to withdraw or modify its approval or recommendation of this Agreement or the Merger in a manner adverse to Acquiror; or (ii) to approve or recommend or enter into an agreement with respect to an Acquisition Proposal; if,

in each such case, (A) an Acquisition Proposal is publicly proposed, publicly disclosed or communicated to the Company and (B) the Board of Directors of the Company determines, based on the advice of the Company's outside legal counsel, that such action is required in order to comply with its fiduciary duties to the stockholders of the Company. No action by the Board of Directors of the Company permitted by the preceding sentence (each, a "Permitted Action") shall constitute a breach of this Agreement by the Company, provided that such Permitted Action shall give rise to the rights of Acquiror set forth in Section 9.3.

- 7.9 Indemnification; Directors' and Officers' Insurance. (a) Until the Effective Time the Company shall, and from and after the Effective Time, Acquiror shall, indemnify, defend and hold harmless each person who is now, or has been at any time prior to the date hereof or who becomes prior to the Effective Time, an officer or director of the Company or any of its Subsidiaries (the "Indemnified Parties") against all losses, claims, damages, costs, expenses (including reasonable attorneys' fees and expenses), liabilities or judgments or amounts that are paid in settlement of or in connection with any threatened or actual claim, action, suit, proceeding or investigation based in whole or substantially on, or arising in whole or substantially out of the fact that such person is or was a director or officer of the Company or any of its Subsidiaries, whether pertaining to any matter existing or occurring at or prior to the Effective Time and whether asserted or claimed prior to, or at or after, the Effective Time ("Indemnified Liabilities"), including all Indemnified Liabilities based in whole or substantially on, or arising in whole or substantially out of, or pertaining to this Agreement or the transactions contemplated hereby, in each case to be fullest extent a corporation is permitted under the DGCL to indemnify its own directors or officers as the case may be (and the Company or the Surviving Corporation, as the case may be, will pay expenses in advance of the final disposition of any such action or proceeding to each Indemnified Party to the fullest extent permitted by law).
- (b) The Acquiror shall and shall cause the Subsidiaries of the Surviving Corporation to keep in effect provisions in their respective Certificates of Incorporation and By-laws providing for exculpation for director, officer and employee liability and such corporation's indemnification of the Indemnified Parties to the fullest extent permitted under the DGCL, which provisions shall not be amended, repealed or otherwise modified for a period of six years after the Effective Time in any manner that would adversely affect the rights thereunder of individuals who at any time prior to the Effective Time were directors, officers or employees of the Company in respect of actions or omissions occurring at or prior to the Effective Time (including, without limitation, the transactions contemplated by this Agreement), unless such modification is required by law.
- (c) For a period of six years after the Effective Time, Acquiror shall cause to be maintained in effect the current policies of directors' and officers' liability insurance maintained by the Company and its Subsidiaries (provided that Acquiror may substitute therefor policies of at least the same coverage and amounts containing terms and conditions that are no less advantageous in any material respect to the Indemnified Parties) with respect to matters arising before the Effective Time, provided that Acquiror shall not be required to pay an annual premium for such insurance in excess of 150% of the last annual premium paid by the Company prior to the date hereof, but in such case shall purchase as much coverage as possible for such amount.
- (d) After the Effective Time, any Indemnified Party wishing to claim indemnification under this Section, upon learning of any such action, suit, claim, proceeding or investigation, shall notify Acquiror within 30 days thereof; provided, however, that any failure so to notify Acquiror of any obligation to indemnify such Indemnified Party or of any other obligation imposed by this Section shall not affect such obligations except to the extent Acquiror is materially prejudiced thereby. Acquiror shall be entitled to assume the defense of any such action, suit, claim, proceeding or investigation with counsel of its choice (who shall be reasonably acceptable to the Indemnified Party), unless there is a conflict between the positions of Acquiror, on the one hand, and the Indemnified Party, on the other, in which event the Indemnified Party, together with all other similarly situated Indemnified Parties in the same proceeding as a group, may retain one law firm and one local counsel to represent them with respect to such matter, the cost of which shall be borne by Acquiror. Neither Acquiror, on the one hand, nor any Indemnified Party, on the other hand, may settle any such action, suit, claim, proceeding or investigation without the prior written consent of the other party, which consent shall not be unreasonably withheld or delayed.

- (e) The provisions of this Section 7.9 are intended to be for the benefit of, and shall be enforceable by, each Indemnified Party, his or her heirs and his or her personal representatives and shall be binding on all successors and assigns of Acquiror and the Company.
- 7.10 NYSE Listing. Acquiror shall use its reasonable efforts to cause the shares of Acquiror Common Stock and New PRIDES to be issued in the Merger to be approved for listing on the NYSE, subject to notice of official issuance, prior to the Effective Time.
- 7.11 Affiliates. Prior to the Closing Date, the Company shall deliver to Acquiror a letter identifying all persons who are, at the time this Agreement is submitted for approval to the stockholders of the Company, "affiliates" of the Company for purposes of Rule 145 under the Securities Act. The Company shall use its best efforts to cause each such person to deliver to Acquiror on or prior to the Closing Date a written agreement substantially in the form attached as Exhibit F.
- 7.12 Rights Agreement. The Board of Directors of the Company shall take all action necessary to render the Company Rights or any similar instrument inapplicable to the Merger and to effect the conversion of the Company Rights as of the Effective Date in accordance with Section 3.1(c) hereof.
- 7.13 Employee Benefits. (a) Acquiror shall or shall cause the Surviving Corporation to maintain in effect employee benefit plans and arrangements which provide benefits which have a value which is substantially comparable, in the aggregate, to the benefits provided by the Employee Benefit Plans (not taking into account the value of any benefits under any such plans which are equity based) for a period of one year after the Effective Time.
- (b) Acquiror shall honor all employment, severance and termination agreements (including change in control provisions) of the employees of the Company and its Subsidiaries.
- (c) For purposes of determining eligibility to participate and vesting, but not accrual or entitlement to benefits (other than severance benefit accrual) where length of service is relevant under any employee benefit plan or arrangement of Acquiror or the Surviving Corporation, employees of the Company and its Subsidiaries as of the Effective Time shall receive service credit for service with the Company and any of its Subsidiaries to the same extent such service was grarted under the Employee Benefit Plans.
- (d) The provisions of Section 7..3(b) are intended to be for the benefit of and shall be enforceable by each applicable employee and his heirs and personal representatives and shall be binding on all successors and assigns of Acquiror, the Surviving Corporation and its Subsidiaries.
- (e) In the event the Surviving Corporation terminates any Employee Benefit Plan which provides pension, profit sharing or non-qualified deferred compensation benefits, the Acquiror shall use its reasonable best efforts to provide the participants and beneficiaries under any such plan with the means to maintain and continue tax deferred treatment of such benefits, other than through the use of an individual retirement account.
- 7.14 Tax Treatment. Each of Acquiror and the Company shall use its best efforts to cause the Merger to qualify as a reorganization under the provisions of Sections 368(a) of the Code and use its reasonable best efforts to obtain the opinions of counsel referred to in Sections 8.2(c) and 8.3(c).
- 7.15 Transfer Taxes. The Company and Acquiror shall cooperate in the preparation, execution and filing of all returns, questionnaires, applications or other documents regarding any real property transfer or gains, sales, use, transfer and stamp taxes, any transfer, recording, registration and other fees and any similar taxes that become payable in connection with the transactions contemplated by this Agreement ("Transfer Taxes"). The Company shall pay or cause to be paid any such Transfer Taxes.
- 7.16 Subsidiary Merger. Immediately prior to the Effective Time, the Company will effect the merger of Bally's Casino, Inc., all of the common stock of which is owned by the Company, with and into the Company

(the "Subsidiary Merger"), with the Company as the surviving corporation, pursuant to the terms of a merger agreement which shall be in form and substance reasonably satisfactory to Acquiror. At the effective time of the Subsidiary Merger (the "Subsidiary Effective Time"), all outstanding shares of Bally's Casino, Inc. common stock shall be cancelled and all outstanding shares of Subsidiary Preferred Stock shall, by virtue of the merger and without any action on the part of the holder thereof, be converted into the right to receive a number of fully paid and non-assessable shares of Company Common Stock (rounded-up to the nearest whole number) equal to the sum of: (i) the number of shares of Company Common Stock exchangeable for the Subsidiary Preferred Stock pursuant to the present terms of the Subsidiary Preferred Stock, and (ii) the number of shares of Company Common Stock equal to the quotient of (A) the product of (1) the average of the last sale prices of the common stock of Bally Total Fitness Holding Corporation ("Bally Fitness") quoted on the Nasdaq National Market (as reported by the Wall Street Journal) for the thirty consecutive trading days ending on the third trading day prior to the Subsidiary Effective Time, and (2) the number of shares of common stock of Bally Fitness transferable pursuant to the present terms of the Subsidiary Preferred Stock upon exchange of such Subsidiary Preferred Stock immediately prior to the Subsidiary Effective Time, divided by (B) the average of the last sale prices of the Company Common Stock quoted on the NYSE for the thirty consecutive trading days ending on the third trading day prior to the Subsidiary Effective Time. All shares of Company Common Stock issuable in the Subsidiary Merger shall be deemed issued and outstanding immediately prior to the Effective Time.

#### ARTICLE VIII

## CONDITIONS PRECEDENT

- 8.1 Conditions to Each Party's Obligation to Effect the Merger. The respective obligation of each party to effect the Merger shall be subject to the satisfaction or waiver on or prior to the Closing Date of the following conditions:
  - (a) Stockholder Approvals. The Company shall have obtained the Company Stockholder Approval and Acquirer shall have obtained the Acquirer Stockholder Approval.
  - (b) HSR Act. The applicable waiting period (and any extension thereof) applicable to the Merger under the HSR Act shall have expired or been earlier terminated.
  - (c) No Injunctions or Restraints. No statute, rule, regulation, decree, preliminary or permanent injunction, temporary restraining order or other order of any nature of any court or Governmental Authority shall be in effect that restrains, prevents or materially changes the transactions contemplated hereby; provided, however, that in the case of a decree, injunction or other order, the party invoking this condition shall have used reasonable efforts to prevent the entry of any such injunction or other order and to appeal as promptly as possible any decree, injunction or other order.
  - (d) Form S-4. The Form S-4 shall have been declared effective under the Securities Act and shall not be the subject of any stop order or proceedings seeking a stop order, and any material "blue sky" and other state securities laws applicable to the issuance of the Acquiror Common Stock and New PRIDES pursuant to this Agreement shall have been complied with.
  - (e) NYSE Listing. The shares of Acquiror Common Stock and New PRIDES issuable to the Company's stockholders in the Merger shall have been approved for listing on the NYSE, subject to official notice of issuance.
- 8.2 Conditions of Obligations of Acquiror. The obligations of Acquiror to effect the Merger are further subject to the satisfaction of the following conditions, any or all of which may be waived on or prior to the Closing Date in whole or in part by Acquiror:
  - (a) Representations and Warranties. The representations and warranties of the Company shall be true and correct in all material respects, at and as of the Closing Date, except for changes permitted or contemplated by this Agreement and except to the extent that any representation or warranty is expressly

A-32

made as of a specified date, in which case such representation or warranty shall be true and correct only as of such date. Acquiror shall have received a certificate from the Company dated the Closing Date signed on behalf of the Company by an authorized officer of the Company certifying to the fulfillment of this condition.

- (b) Agreements. The Company shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to or at the Closing Date and shall have complied or be in compliance in all material respects with any agreement or covenant of the Company to be performed by it under this Agreement at or prior to the Closing Date, and Acquiror shall have received a certificate from the Company dated the Closing Date signed on behalf of the Company by an authorized officer of the Company certifying to the fulfillment of this condition.
- (c) Tax Opinion. Acquirer shall have received an opinion of Latham & Watkins, dated the Closing Date, to the effect that for federal income tax purposes (i) the Merger will constitute a reorganization within the meaning of Section 368(a) of the Code; (ii) each of Acquirer and the Company will be a party to the reorganization within the meaning of Section 368(b) of the Code; and (iii) no gain or loss will be recognized by the Company or Acquirer as a result of the Merger, and substantially in the form of Exhibit A. In rendering such opinion, Latham & Watkins shall receive and may rely upon representations contained in certificates of the Company, Acquirer, and certain stockholders of the Company substantially in the forms of Exhibits C, D, and E.
- (d) Letters from Affiliates. Acquiror shall have received from each person in the letter referred to in Section 7.11 an executed copy of an agreement substantially in the form of Exhibit F.
- (e) Consents. All necessary approvals or authorizations of any Governmental Authority required or necessary under applicable Gaming Laws in connection with the Merger shall have been obtained.
- (f) Litigation. There shall not have entered any order by any Governmental Authority in any suit, action or proceeding, which (i) restrains or prohibits the Merger or any of the other transactions contemplated by this Agreement, (ii) requires the payment of damages by Acquiror in connection with the Merger which damages are material to the Acquiror and its Subsidiaries taken as a whole (determined after giving effect to the Merger), (iii) prohibits or limits the ownership or operation by Acquiror and its Subsidiaries of, or compels Acquiror or any of its Subsidiaries to dispose of or hold separate, any business or assets which are material to Acquiror and its Subsidiaries taken as a whole, in each case as a result of the Merger or any of the other transactions contemplated by this Agreement, (iv) imposes limitations on the ability of Acquirer to acquire or hold, or exercise full rights of ownership of, shares of capital stock of the Subsidiaries of the Company, which limitations would have a Material Adverse Effect with respect to Acquiror (determined after giving effect to the Merger).
- (g) Company Rights. The Company Rights shall not have become nonredeemable, exercisable or triggered pursuant to the terms of the Company Rights Agreement.
- 8.3 Conditions of Obligation of the Company. The obligation of the Company to effect the Merger is further subject to the satisfaction of the following conditions, any or all of which may be waived on or prior to the Closing Date in whole or in part by the Company:
  - (a) Representations and Warranties. The representations and warranties of Acquiror made hereunder shall be true and correct in all material respects at and as of the Closing Date, except for changes permitted or contemplated by this Agreement and except to the extent that any representation or warranty is expressly made as of a specified date, in which case such representation or warranty shall be true and correct only as of such date. The Company shall have received a certificate from Acquiror dated the Closing Date signed on behalf of Acquiror by an authorized officer of Acquiror certifying to the fulfillment of this condition.
  - (b) Agreements. Acquirer shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to or at the Closing Date and shall have complied or be in compliance in all material respect; with any applicable agreement or covenant of Acquirer to be performed

A-33

by Acquiror under this Agreement at or prior to the Closing Date, and the Company shall have received the certificates from the Acquiror dated the Closing Date signed on behalf of Acquiror by an authorized officer of Acquiror certifying to the fulfillment of this condition.

- (c) Litigation. There shall not have entered any order by any Governmental Authority in any suit, action or proceeding, which (i) requires the payment of damages by Acquiror which would have a Material Adverse Effect with respect to Acquiror (determined after giving effect to the Merger), (ii) prohibits or materially limits the ownership or operation by Acquiror or any of its Subsidiaries of, or compels Acquiror or any of its Subsidiaries to dispose of or hold separate, any business or assets which are material to Acquiror and its Subsidiaries taken as a whole (determined after giving effect to the Merger), in each case as a result of the Merger or any of the other transactions contemplated by this Agreement, or (iii) which otherwise is reasonably likely to have a Material Adverse Effect with respect to Acquiror (determined after giving effect to the Merger).
- (d) Tax Opinion. The Company shall have received an opinion of Weil, Gotshal & Manges LLP, dated the Closing Date, to the effect that for federal income tax purposes (i) the Merger will constitute a reorganization within the meaning of Section 368(a) of the Code; (ii) each of Acquiror and the Company will be a party to the reorganization within the meaning of Section 368(b) of the Code; and (iii) no gain or loss will be recognized by a stockholder of the Company as a result of the Merger except (x) with respect to cash received by such stockholder in lieu of fractional shares and (y) if a stockholder receives any Cash Consideration, gain, if any, realized by such stockholder pursuant to the Merger will be recognized, but not in excess of any Cash Consideration received, and substantially in the form of Exhibit B. In rendering such opinion, Weil, Gotshal & Manges LLP shall receive and may rely upon representations contained in certificates of the Company, Acquiror and certain stockholders of the Company substantially in the forms of Exhibits C, D and E.

#### ARTICLE IX

## TERMINATION AND AMENDMENT

- 9.1 Termination. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the Company Stockholder Approval or Acquiror Stockholder Approval have been obtained:
  - (a) by mutual written consent of the Company, on the one hand, and Acquiror, on the other hand, or by mutual action of their respective boards of directors;
  - (b) by Acquiror, if any of the conditions set forth in Section 8.2 shall have become incapable of fulfillment (other than as a result of any breach by Acquiror of the terms of this Agreement) and shall not have been waived by Acquiror;
  - (c) by the Company, if any of the conditions set forth in Section 8.3 shall have become incapable of fulfillment (other than as a result of any breach by the Company of the terms of this Agreement), and shall not have been waived by the Company;
  - (d) by either the Company or Acquiror, if the Merger shall not have been consummated on or before the earlier of (e) the 90th day following the mailing of the Proxy Statement/Prospectus and (ii) March 31, 1997 (the "Termination Date"), unless the failure to consummate the Merger is the result of a willful and material breach of this Agreement by the party seeking to terminate this Agreement;
  - (e) by either the Company or Acquiror if any of the conditions set forth in Section 8.1 shall have become incapable of fulfillment (other than as a result of any breach by the party seeking to terminate the Agreement) and shall not have been waived in accordance with the terms of this Agreement;
    - (f) by the Company, pursuant to Section 3.1(c)(i);

A-34

- (g) by Acquiror, if the Board of Directors of the Company shall have taken any Permitted Action in accordance with the provisions of Section 7.8(d); or
- (h) by the Company if (i) the Board of Directors pursuant to Section 7.8(d) withdraws or modifies its approval or recommendation of this Agreement or (ii) the Company enters into a definitive agreement providing for the implementation of an Acquisition Proposal in accordance with the provisions of Section 7.8.
- 9.2 Effect of Termination. In the event of termination by the Company or Acquiror pursuant to Section 9.1, written notice thereof shall promptly be given to the other parties and, except as otherwise provided herein, the transactions contemplated by this Agreement shall be terminated and become void and have no effect, without further action by any party, other than the provisions of the last sentence of Section 6.3 and Sections 9.2 and 9.3 and Article X. Nothing in this Section 9.2 shall be deemed to release any party from any liability for any willful and material breach by such party of the terms and provisions of this Agreement.
- 9.3 Fees and Expenses. (a) In order to induce Acquiror to, among other things, enter into this Agreement, the Company agrees to pay Acquiror the fees and expenses set forth in Sections 9.3(b) and 9.3(c).
- (b) Unless there shall have been a material misrepresentation by or material breach of any material obligation of Acquiror hereunder which would entitle the Company to terminate this Agreement under Section 9.1, if:
  - (i) an Acquisition Proposal is commenced, publicly proposed, publicly disclosed or communicated to the Company after the date of this Agreement, and (A) the Board of Directors of the Company takes any Permitted Action under Section 7.8(d) and (B) this Agreement is terminated by Acquiror pursuant to Section 9.1(g) following the Company's taking a Permitted Action or by the Company pursuant to Section 9.1(h); or
  - (ii) an Acquisition Proposal is publicly commenced, proposed or disclosed after the date of this Agreement and prior to the date of the Company Stockholders Meeting and this Agreement is terminated pursuant to Section 9.1(e) following the rejection of this Agreement at the Company Stockholders Meeting; or
  - (iii) this Agreement is terminated by Acquiror pursuant to Section 9.1(b) based on the failure of the condition set forth in Section 8.2(g);

and within one year following any such termination, an Acquisition Proposal is consummated or the Company enters into an agreement with respect thereto, then the Company shall pay to Acquiror within five business days following such occurrence a fee (the "Topping Fee") of \$50 million in cash; provided, however, that in no event shall the Company be obligated to pay more than one such fee with respect to all such occurrences.

- (c) If a Topping Fee is payable to Acquiror pursuant to Section 9.3(b), then the Company shall also reimburse Acquiror (not later than five business days after submission of statements therefor) for all documented out-of-pocket (not including allocation of overhead) fees and expenses ("Expenses") actually incurred by it prior to such termination in connection with the proposed Merger and the consummation of all transactions contemplated by this Agreement (including fees and expenses of counsel, investment banking firms, accountants, experts and consultants); provided, that the aggregate amount of Expenses required to be reimbursed pursuant to this Section shall not exceed S5 million.
- 9.4 Amendment. Subject to Applicable Law, this Agreement may be amended, modified or supplemented only by written agreement of Acquiror and the Company at any time prior to the Effective Date with respect to any of the terms contained herein; provided, however, that, after receipt of the Company Stockholder Approval, no such amendment or modification shall reduce the amount or change the form of consideration to be delivered to the stockholders of the Company

9.5 Extension; Waiver. At any time prior to the Effective Time, the parties hereto, by action taken or authorized by their respective boards of directors, may, to the extent legally allowed: (i) extend the time for the performance of any of the obligations or other acts of the other parties hereto; (ii) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto; and (iii) waive compliance with any of the agreements or conditions contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of such party. The failure of any party hereto to assert any of its rights hereunder shall not constitute a waiver of such rights.

## ARTICLE X

## GENERAL PROVISIONS

- 10.1 Effectiveness of Representations, Warranties and Agreements. The representations, warranties and agreements in this Agreement shall terminate at the Effective Time or upon the termination of this Agreement pursuant to Article IX except that the agreements set forth in Articles II and III and Sections 7.9 and 7.13 shall survive the Effective Time and those set forth in Sections 9.2, 9.3 and Article X hereof shall survive termination.
- 10.2 Expenses. Except as otherwise provided herein, including in Section 9.3, each of the parties hereto shall pay the fees and expenses of its respective counsel, accountants and other experts and shall pay all other costs and expenses incurred by it in connection with the negotiation, preparation and execution of this Agreement and the consummation of the transactions contemplated hereby.
- 10.3 Governing Law. This Agreement shall be governed by, and construed in accordance with, the laws of the State of New York without reference to choice of law principles, including all matters of construction, validity and performance (except to the extent that the provisions of the DGCL shall be mandatorily applicable to the Merger or this Agreement).
- 10.4 Notices. Notices, requests, permissions, waivers, and other communications hereunder shall be in writing and shall be deemed to have been duly given if signed by the respective persons giving them (in the case of any corporation the signature shall be by an officer thereof) and delivered by hand, deposited in the United States mail (registered or certified, return receipt requested), properly addressed and postage prepaid, or delivered by telecopy:

If to the Company, to:

Bally Entertainment Corporation 8700 West Bryn Mawr Avenue Chicago, Illinois 60631 Attention: James S. Montana, Jr., Esq.

Facs mile: (312) 399-0168

with a copy to:

Weil. Gotshal & Manges LLP 767 Fifth Avenue New York, NY 10153 Attention: Dennis J. Block, Esq.

Facsimile: (212) 310-8007

A-36

and:

If to Acquiror, to:

Hilton Hotels Corporation
9336 Civic Center Drive
Beverly Hills, CA 90210
Attention: William C. Lebo, Jr., Esq.

Facsimile: (310) 205-7677

with a copy to:

Latham & Watkins
1001 Pennsylvania Avenue, N.W.
Washington, D.C. 20004
Attention: Bruce E. Rosenblum, Esq.

Facsimile: (202) 637-2201

Such names and addresses may be changed by notice given in accordance with this Section 10.4.

10.5 Entire Agreement. This Agreement (including the Company Disclosure Schedule, the Acquiror Disclosure Schedule and the Exhibits attached hereto, all of which are a part hereof) and the Confidentiality Agreement contain the entire understanding of the parties hereto and thereto with respect to the subject matter contained herein and therein, supersede and cancel all prior agreements, negotiations, correspondence, undertakings and communications of the parties, oral or written, respecting such subject matter. There are no restrictions, promises representations, warranties, agreements or undertakings of any party hereto with respect to the transactions contemplated by this Agreement other than those set forth herein or made hereunder.

10.6 Disclosure Schedule. The Disclosure Schedule, dated the date hereof, delivered by the Company to Acquiror (the "Company Disclosure Schedule") and the Disclosure Schedule, dated the date hereof, delivered by Acquiror to the Company (the "Acquiror Disclosure Schedule") are incorporated into this Agreement by reference and made a part hereof. Nothing disclosed in the Company Disclosure Schedule or the Acquiror Disclosure Schedule shall be deemed to be an admission that such matters are material or are required to be disclosed herein. Certain immaterial items or items that are not entirely responsive to the information required in a Schedule may be included in various Schedules as further clarification or assistance to the parties in understanding the business and operations of the parties or in consummating the transactions contemplated herein

- 10.7 Headings; References. The article, section and paragraph headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. All references herein to "Articles", "Sections" or "Exhibits" shall be deemed to be references to Articles or Sections hereof or Exhibits hereto unless otherwise indicated.
- 10.8 Counterparts. This Agreement may be executed in one or more counterparts and each counterpart shall be deemed to be an original, but all of which shall constitute one and the same original.
- 10.9 Parties in Interest; Assignment. Neither this Agreement nor any of the rights, interest or obligations hereunder shall be assigned by any of the parties hereto without the prior written consent of the other parties. Subject to the preceding sentence this agreement shall inure to the benefit of and be binding upon the Company and Acquiror and shall inure to the sole benefit of the Company and Acquiror and their respective successors and permitted assigns. Except as set forth in Sections 7.9 and 7.13 nothing in this Agreement, express or implied, is intended to confer upon any other Person any rights or remedies under or by reason of this Agreement.
- 10.10 Severability; Enforcement. Except to the extent that the application of this Section 10.10 would have a Material Adverse Effect with respect to Acquiror or the Company, the invalidity of any portion hereof shall

not affect the validity, force or effect of the remaining portions hereof. If it is ever held that any restriction hereunder is too broad to permit enforcement of such restriction to its fullest extent, each party agrees that a court of competent jurisdiction may enforce such restriction to the maximum extent permitted by law, and each party hereby consents and agrees that such scope may be judicially modified accordingly in any proceeding brought to enforce such restriction.

10.11 Specific Performance. The parties hereto agree that the remedy at law for any breach of this Agreement will be inadequate and that any party by whom this Agreement is enforceable shall be entitled to specific performance in addition to any other appropriate relief or remedy. Such party may, in its sole discretion, apply to a court of competent jurisdiction for specific performance or injunctive or such other relief as such court may deem just and proper in order to enforce this Agreement or prevent any violation hereof and, to the extent permitted by Applicable Law, each party waives any objection to the imposition of such relief.

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

# HILTON HOTELS CORPORATION

By: /s/ STEPHEN F. BOLLENBACH

Name: Stephen F. Bollenbach Title: President and

Chief Executive Officer

# BALLY ENTERTAINMENT CORPORATION

By: /s/ Arthur M. Goldberg

Name: Arthur M. Goldberg
Title: Chairman of the Board,
President and Chief
Executive Officer

# WELSH & KATZ, LTD.

Attornerys at Law

120 South Riverside Plaza - 22nd Floor Chicago, Illinois 60606

> TELEPHONE (312) 655-1500 FACSIMILE (312) 655-1501

JOSEPH R. MARCUS
GERALD S. SCHUR
GERALD T. SHEKLETON
JAMES A. SCHEER
DANIEL R. CHERRY
ROBERT B. BREISBLATT
JAMES P. WHITE
R. MARK HALLIGAN
HARTWELL P. MORSE, III
EDWARD P. GAMSON, PH.D.
KARA E.F. CENAR
KATHLEEN A. RHEINTGEN
THOMAS W. TOLPIN\*
ELLIOTT C. BANKENDORF
\*ALSO ADMITTED IN DISTRICT OF CCL

A. SIDNEY KATZ\*

ERIC C. COHEN

RICHARD L. WOOD\*

JEROLD B. SCHNAYER



JOHN L. AMBROGI
JULIE A. KATZ
ERIC D. COHEN
JON P. CHRISTENSEN
WALTER J. KAWULA, JR.
LEONARD FRIEDMAN
STEVEN E. FELDMAN
IK HYUN SEO
PHILIP D. SEGREST, JR.
JEFFREY W. SALMON
MITCHELL J. WEINSTEIN
SHANNON L. NEBOLSKY, PH.D.
ELIZABETH D. McGOOGAN

OF COUNSEL
DONALD L. WELSH
JAMES W. CLEMENT
LAURIE A. HAYNIE

WASHINGTON OFFICE CRYSTAL PLAZA ONE • SUITE BOB 2001 JEFFERSON DAVIS HIGHWAY ARLINGTON, VIRGINIA 22202-3603 TELEPHONE (703) 415-4777

Box Assignments Assistant Commissioner for Trademarks 2900 Crystal Drive Arlington, VA 22202-3513

Re: Recording of Merger between Hilton Hotels Corporation

and Bally Entertainment Corporation

Our File No. 2785/57875

Sir:

Enclosed, for recording, is a Trademark Assignment Cover Sheet and Exhibit A, a list of 70 Bally Entertainment Corporation registrations and applications. Also enclosed is a copy of the Agreement and Plan of Merger between Hilton Hotels Corporation and Bally Entertainment Corporation, whereby Bally Entertainment Corporation merged into Hilton Hotels Corporation.

Please cause these documents to be recorded and indexed against the trademark and service mark registrations and applications listed in Exhibits A attached, as soon as possible. When this is complete, please return the recorded document to the undersigned. Our check in the amount of \$1765.00 is enclosed to cover the recording fee.

The Commissioner is hereby authorized to charge any additional fees which may be required, including, if necessary, the filing fee if the above-referenced check is in the wrong amount, unsigned, postdated, or otherwise improper or informal or missing, or credit any overpayment to Deposit Account No. 23-0920. A duplicate copy of this sheet is enclosed.

Respectfully,

WELSH & KATZ, LTD.

By Mili

JAK:cb Enclosures

Richard Barrier, Esq.

A. Sidney Katz, Esq.

# WELSH & KATZ, LTD.

Attorneys at Law

120 South Riverside Plaza - 22nd Floor Chicago, Illinois 60606

> TELEPHONE (312) 655-1500 FACSIMILE (312) 655-1501

OF COUNSEL LAURIE A. HAYNIE JAMES J. MYRICK

ERIC D. COHEN

WALTER J. KAWULA, JR.

LEONARD FRIEDMAN

STEVEN E. FELDMAN PHILIP D. SEGREST, JR.

JEFFREY W. SALMON

RICHARD J. GURAK

J. ARON CARNAHAN

RALPH E. KRISHER III

THOMAS L. GEMMELL

ERIK B. FLOM, PH.D.

MICHAEL A. BONDI

LOUISE T. WALSH

JOSEPH E. CWIK

SCOTT M. GETTLESON

MITCHELL J. WEINSTEIN

ELIZABETH D. McGOOGAN

SHANNON L. NEBOLSKY, Ph.D.

WASHINGTON OFFICE CRYSTAL PLAZA ONE - SUITE 206 2001 JEFFERSON DAVIS HIGHWAY ARLINGTON, VIRGINIA 22202-3603 TELEPHONE (703) 415-4777

CHARLES R. KRIKORIAN, PH.D.

October 21, 1999

Via Express Mail No. EL215072208US

DONALD L. WELSH 11925-1998

A SIDNEY KATZ\* RICHARD L. WOOD\*

ERIC C. COHEN

JEROLD B. SCHNAYER

GERALD T. SHEKLETON

ROBERT B. BREISBLATT

HARTWELL P. MORSE, III

EDWARD P. GAMSON, PH.D.

KATHLEEN A, RHEINTGEN

ELLIOTT C. BANKENDORF

RICHARD W. McLAREN, JR.

JOSEPH R. MARCUS GERALD S. SCHUR

JAMES A. SCHEER

DANIEL R. CHERRY

R. MARK HALLIGAN

KARA E.F. CENAR

THOMAS W. TOLPIN\*

JOHN L. AMBROGI

JON P. CHRISTENSEN

JULIE A. KATZ

JAMES P. WHITE

\* ALSO ADMITTED IN DISTRICT OF COLUMBIA

**BOX ASSIGNMENTS** Commissioner of Patents and Trademarks Washington, D.C. 20231

Recordal of Assignment Trademark Registration No. 1,773,050

Our File No.: 2782/60547

Sir:

Enclosed for recording is an Assignment from Bally Entertainment Corporation to Hilton Hotels Corporation and Recordation Form Cover Sheet therefor. Please return the recorded document to the undersigned. A check in the amount of \$40.00 is enclosed to cover the recording fee.

Also, please note that Assignment of the above-noted Registration was supposed to have been recorded previously at Reel/Frame 1590/0169. However, the Registration. No. is stated incorrectly. Registration No. 1,733,050 for the mark ELK is improperly set forth; is not part of the assignment between Bally Entertainment Corporation and Hilton Hotels Corporation because Bally is not the owner of the ELK registration; and, should be removed from the Assignment Branch record. The Registration No. should be 1,773,050 for the mark VERTICAL CLUB and Design. A copy of the previous Notice of Recordation is enclosed for your ease of reference.

The Commissioner is hereby authorized to charge any additional fees which may be required, including, if necessary, the filing fee if the above-referenced check is in the wrong amount, unsigned, postdated, or otherwise to Deposit Account No. 23-0920. A copy of this sheet is enclosed.

Commissioner of Patents and Trademarks October 21, 1999 Page 2

If you have any additional questions, please contact the undersigned.

Respectfully,

WELSH & KATZ, LTD

JAK:lc Enclosures

cc: A. Sidney Katz, Esq. (w/o encls.)



UNITED STATES DEPARTMENT OF COMMERCE Patent and Trademark Office

ASSISTANT SECRETARY AND COMMISSIONER OF PATENTS AND TRADEMARKS Washington, D.C. 20231

JULY 07, 1997

PTAS

WELSH & KATZ, ESQ.

JULIE A. KATZ, ESQ.

120 S. RIVERSIDE PLAZA 22ND FLOOR

CHICAGO, IL 60606



UNITED STATES PATENT AND TRADEMARK OFFICE NOTICE OF RECORDATION OF ASSIGNMENT DOCUMENT

THE ENCLOSED DOCUMENT HAS BEEN RECORDED BY THE ASSIGNMENT DIVISION OF THE U.S. PATENT AND TRADEMARK OFFICE. A COMPLETE MICROFILM COPY IS AVAILABLE AT THE ASSIGNMENT SEARCH ROOM ON THE REEL AND FRAME NUMBER REFERENCED BELOW.

PLEASE REVIEW ALL INFORMATION CONTAINED ON THIS NOTICE. THE INFORMATION CONTAINED ON THIS RECORDATION NOTICE REFLECTS THE DATA PRESENT IN THE PATENT AND TRADEMARK ASSIGNMENT SYSTEM. IF YOU SHOULD FIND ANY ERRORS OR HAVE QUESTIONS CONCERNING THIS NOTICE, YOU MAY CONTACT THE EMPLOYEE WHOSE NAME APPEARS ON THIS NOTICE AT 703-308-9723. PLEASE SEND REQUEST FOR CORRECTION TO: U.S. PATENT AND TRADEMARK OFFICE, ASSIGNMENT DIVISION, BOX ASSIGNMENTS, NORTH TOWER BUILDING, SUITE 10C35, WASHINGTON, D.C. 20231.

RECORDATION DATE: 05/12/1997

REEL/FRAME: 1590/0169 NUMBER OF PAGES: 55

BRIEF: MERGER

ASSIGNOR:

BALLY ENTERTAINMENT CORPORATION

DOC DATE: 06/06/1996 CITIZENSHIP: DELAWARE ENTITY: Corporation

CITIZENSHIP: CALIFORNIA

FILING DATE: 04/16/1991

ISSUE DATE: 09/08/1992

ENTITY: Corporation

ASSIGNEE:

HILTON HOTELS CORPORATION
9336 CIVIC CENTER DRIVE
BEVERLY HILLS CALLEORNIA 90210

BEVERLY HILLS, CALIFORNIA 90210

APPLICATION NUMBER: 74157749 REGISTRATION NUMBER: 1713095

MARK: RAZOR TOOTH SAW

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74157797 FILING DATE: 04/16/1991 REGISTRATION NUMBER: 1711772 ISSUE DATE: 09/01/1992

MARK: SIMON PEARCE

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

TRADEMARK
REEL: 001979 FRAME: 0561

134

1590/0169 PAGE 2

APPLICATION NUMBER: 74162713 FILING DATE: 05/01/1991 REGISTRATION NUMBER: 1748321 ISSUE DATE: 01/26/1993

MARK:

DRAWING TYPE: MISCELLANEOUS DESIGN

APPLICATION NUMBER: 74638246 FILING DATE: 02/24/1995

ISSUE DATE: REGISTRATION NUMBER:

MARK: BALLY TOTAL FITNESS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74640368 FILING DATE: 02/24/1995 REGISTRATION NUMBER: ISSUE DATE:

MARK: BALLY'S PROACTIVE SERVICE

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

FILING DATE: 04/09/1991 APPLICATION NUMBER: 74155687

REGISTRATION NUMBER:

MARK: BALLY

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 74162874 FILING DATE: 05/02/1991 REGISTRATION NUMBER: 1729468 ISSUE DATE: 11/03/1992

MARK: STATUSWEAR

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74165335 FILING DATE: 05/10/1991

REGISTRATION NUMBER: ISSUE DATE:

MARK: SCA-LP

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74169129 FILING DATE: 05/22/1991

ISSUE DATE: REGISTRATION NUMBER:

MARK: FLORIDA GOLFWEEK

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74219597 FILING DATE: 11/07/1991

REGISTRATION NUMBER: ISSUE DATE:

MARK: FAST FIFTIES

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74481961 FILING DATE: 01/21/1994

REGISTRATION NUMBER: ISSUE DATE:

MARK: BALLY

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

1590/0169 PAGE 3

APPLICATION NUMBER: 74608793 FILING DATE: 12/09/1994

REGISTRATION NUMBER: ISSUE DATE:

MARK: VC

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

FILING DATE: 08/31/1995 APPLICATION NUMBER: 74723086

REGISTRATION NUMBER: ISSUE DATE:

MARK: BALLY'S LAKEHOUSE

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74536223 FILING DATE: ISSUE DATE: FILING DATE: 06/10/1994

MARK: BILLY'S

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74541524 FILING DATE: 06/10/1994 REGISTRATION NUMBER: ISSUE DATE:

MARK: BILLY'S

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74711619 FILING DATE: 08/07/1995 REGISTRATION NUMBER: 2048291 ISSUE DATE: 03/25/1997

MARK: BALLY'S SALOON GAMBLING HALL HOTEL

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 741.03801 FILING DATE: 10/09/1990 REGISTRATION NUMBER: 1696363 ISSUE DATE: 06/23/1992

MARK: CAMELOT

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74176918 FILING DATE: 06/17/1991 REGISTRATION NUMBER: 1719947 ISSUE DATE: 09/29/1992

MARK: SAVIC

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 74177000 FILING DATE: 06/17/1991 REGISTRATION NUMBER: 1810502 ISSUE DATE: 12/14/1993

MARK: GOLDDISC COMPACT DIGITAL AUDIO

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74183087 FILING DATE: 07/08/1991

REGISTRATION NUMBER: ISSUE DATE:

MARK: PUDDLES

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74208253

REGISTRATION NUMBER:

FILING DATE: 09/27/1991

ISSUE DATE:

ISSUE DATE:

MARK: TO THINE OWN SELF BE TRUE RARELY HAVE WE

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 74653425 FILING DATE: 03/29/1995

REGISTRATION NUMBER:

MARK: PARIS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74703331 FILING DATE: 07/18/1995 REGISTRATION NUMBER: ISSUE DATE:

REGISTRATION NUMBER:

MARK: PARIS CASINO - RESORT

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74703332 FILING DATE: 07/18/1995 REGISTRATION NUMBER: ISSUE DATE:

REGISTRATION NUMBER:

MARK: PARIS CASINO RESORT

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74006844 FILING DATE: 12/01/1989 REGISTRATION NUMBER: 1623605 ISSUE DATE: 11/20/1990

MARK: BALLY'S TOTAL FITNESS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74421040 FILING DATE: 08/04/1993 REGISTRATION NUMBER: 1854914 ISSUE DATE: 09/20/1994

MARK: BALLY'S

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74421427 FILING DATE: 08/04/1993 REGISTRATION NUMBER: 1870035 ISSUE DATE: 12/27/1994

MARK: BALLY'S

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74503405 FILING DATE: 03/18/1994 REGISTRATION NUMBER: 1922150 ISSUE DATE: 09/26/1995

MARK: WIDE OPEN

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74640369 FILING DATE: 02/24/1995 REGISTRATION NUMBER: 2030899 ISSUE DATE: 01/14/1997

MARK: B

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

## 1590/0169 PAGE 5

APPLICATION NUMBER: 74655244 FILING DATE: 04/03/1995 REGISTRATION NUMBER: 1973721 ISSUE DATE: 05/14/1996

MARK: B BALLY TOTAL FITNESS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 73687446 FILING DATE: 10/02/1987 REGISTRATION NUMBER: 1552562 ISSUE DATE: 08/22/1989

MARK: BALLY

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74689601 FILING DATE: 06/16/1995 REGISTRATION NUMBER: 2020345 FILING DATE: 12/03/1996

MARK: B BALLY TOTAL FITNESS

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74800026 FILING DATE: 12/01/1989 REGISTRATION NUMBER: 1673918 ISSUE DATE: 01/28/1992

MARK: BALLY'S TOTAL FITNESS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 80639034 FILING DATE: REGISTRATION NUMBER: 0639034 ISSUE DATE:

MARK:

DRAWING TYPE:

APPLICATION NUMBER: 73C24101 FILING DATE: 06/13/1974 REGISTRATION NUMBER: 1C12119 ISSUE DATE: 06/03/1975

MARK: BALLY

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74C63579 FILING DATE: 05/29/1990 REGISTRATION NUMBER: 1646697 ISSUE DATE: 06/04/1991

MARK: BALLY

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74104800 FILING DATE: 10/09/1990 REGISTRATION NUMBER: 1666141 ISSUE DATE: 11/26/1991

MARK: VERTICAL CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74155686 FILING DATE: 04/09/1991 REGISTRATION NUMBER: 1909427 ISSUE DATE: 08/01/1995

MARK: BALLY

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74174748 FILING DATE: 06/10/1991 REGISTRATION NUMBER: 1724335 ISSUE DATE: 10/13/1992

MARK: THE VERTICAL CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74210159 FILING DATE: 10/04/1991 REGISTRATION NUMBER: 1715953 ISSUE DATE: 09/15/1992

MARK: VERTICAL CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 73223362 FILING DATE: 07/16/1979 REGISTRATION NUMBER: 1151961 ISSUE DATE: 04/21/1981

MARK: VERTICAL CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 74262099 FILING DATE: 04/03/1992 REGISTRATION NUMBER: 1734972 ISSUE DATE: 11/24/1992

MARK: VERTICAL CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74265904 FILING DATE: 04/14/1992 REGISTRATION NUMBER: 1792139 ISSUE DATE: 09/07/1993

MARK: BALLY CAJUN

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74265906 FILING DATE: 04/14/1992 REGISTRATION NUMBER: 1806687 FILING DATE: 11/23/1993

MARK: BALLY MARDI GRAS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74265907 FILING DATE: 04/14/1992 REGISTRATION NUMBER: 1816368 ISSUE DATE: 01/11/1994

MARK: BALLY MAVERICK

DRAWING TYPE: WORDS, LEGTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74282077 FILING DATE: 06/04/1992 REGISTRATION NUMBER: 1820490 ISSUE DATE: 02/08/1994

MARK: BALLY GAMING

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74234290 FILING DATE: 12/30/1991 REGISTRATION NUMBER: 1733050 FILING DATE: 11/17/1992

MARK: ELK

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 74482460 FILING DATE: 01/21/1994 REGISTRATION NUMBER: 1904615 ISSUE DATE: 07/11/1995

MARK: BALLY'S

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74485541 FILING DATE: 02/02/1994 REGISTRATION NUMBER: 1915383 FILING DATE: 08/29/1995

MARK: BALLY'S

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 74487140 FILING DATE: 02/07/1994 REGISTRATION NUMBER: 2012257 ISSUE DATE: 10/29/1996

MARK: BALLY'S BELLE OF ORLEANS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 71532200 FILING DATE: 08/23/1947 REGISTRATION NUMBER: 0505218 ISSUE DATE: 12/28/1948

MARK: BALLY

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 73619697 FILING DATE: 09/15/1986 REGISTRATION NUMBER: 1440235 ISSUE DATE: 05/19/1987

MARK: 7 HEAVEN

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73620651 FILING DATE: 09/18/1986 REGISTRATION NUMBER: 1439550 ISSUE DATE: 05/12/1987

MARK: 7 HEAVEN DOLLARS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73625038 FILING DATE: 10/10/1986 REGISTRATION NUMBER: 1439557 ISSUE DATE: 05/12/1987

MARK: 777 HEAVEN

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73666811 FILING DATE: 06/12/1987 REGISTRATION NUMBER: 1515843 ISSUE DATE: 12/06/1988

MARK: BALLY'S

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73666822 FILING DATE: 06/15/1987 REGISTRATION NUMBER: 1523910 ISSUE DATE: 02/07/1989

MARK: BALLY'S

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73693699 FILING DATE: 11/05/1987 REGISTRATION NUMBER: 1547119 ISSUE DATE: 07/11/1989

MARK: BALLY SYSTEM 50(0

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 73697657 FILING DATE: 11/25/1987 REGISTRATION NUMBER: 1533712 ISSUE DATE: 04/04/1989

MARK: BALLY'S GRAND

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73697660 FILING DATE: 11/25/1987 REGISTRATION NUMBER: 1543909 ISSUE DATE: 06/13/1989

MARK: BALLY'S GRAND

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 71699197 FILING DATE: 12/01/1955 REGISTRATION NUMBER: (637106 ISSUE DATE: 11/13/1956

MARK: PIN-POOL

DRAWING TYPE: STYLIZED WORDS, LETTERS, OR NUMBERS

APPLICATION NUMBER: 73809446 FILING DATE: 06/27/1989 REGISTRATION NUMBER: 1581959 ISSUE DATE: 02/06/1990

MARK: BALLY'S

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 73051092 FILING DATE: 05/01/1975 REGISTRATION NUMBER: 1066701 ISSUE DATE: 05/31/1977

MARK: YOUR FAMILY GUIDE

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74227176 FILING DATE: 12/04/1991 REGISTRATION NUMBER: 1847026 ISSUE DATE: 07/26/1994

MARK: BIG BUCKS

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74535501 FILING DATE: 06/09/1994 REGISTRATION NUMBER: 1955688 ISSUE DATE: 02/13/1996

MARK: BALLY BOB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74538625 FILING DATE: 06/16/1994 REGISTRATION NUMBER: 1955699 ISSUE DATE: 02/13/1996

MARK: BALLY'S SALOON & GAMBLING HALL

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

## 1590/0169 PAGE 9

APPLICATION NUMBER: 74543447 FILING DATE: 06/28/1994 REGISTRATION NUMBER: 1975755 ISSUE DATE: 05/28/1996

MARK:

DRAWING TYPE: MISCELLANEOUS DESIGN

APPLICATION NUMBER: 74655246 FILING DATE: 04/03/1995 REGISTRATION NUMBER: 1973722 ISSUE DATE: 05/14/1996

MARK: BALLY'S CASINO LAKESHORE RESORT

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74659487 FILING DATE: 04/03/1995 REGISTRATION NUMBER: 1982255 ISSUE DATE: 06/25/1996

MARK: BALLY'S CASINO LAKESHORE RESORT NEW ORLE

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

APPLICATION NUMBER: 74666975 FILING DATE: 04/27/1995 REGISTRATION NUMBER: 2032928 ISSUE DATE: 01/21/1997

MARK: BALLY'S COMMODORE CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS IN TYPED FORM

APPLICATION NUMBER: 74689578 FILING DATE: 06/16/1995 REGISTRATION NUMBER: 2034472 ISSUE DATE: 01/28/1997

MARK: BALLY'S COMMODORE CLUB

DRAWING TYPE: WORDS, LETTERS, OR NUMBERS AND DESIGN

DOROTHY RILEY, EXAMINER ASSIGNMENT DIVISION OFFICE OF PUBLIC RECORDS

**RECORDED: 10/21/1999**